UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): May 16, 2016

ReWalk Robotics Ltd.			
(Exact name of registrant as specified in its charter)			
Israel	001-36612	N/A	
(State or Other Jurisdiction of Incorporation)	(Commission File Number)	(IRS Employer Identification No.)	
3 Hatnufa St., Floor 6, Yokneam Ilit, Israel		2069203	
(Address of principal executive offices)		(Zip Code)	
Registrant's telephone number, including area code:	+972.4.959.0123		
	Not applicable		
(Former	name or former address, if changed since last r	report)	
Check the appropriate box below if the Form 8-K following provisions (<u>see</u> General Instruction A.2. below	-	ne filing obligation of the registrant under any of the	
☐ Written communications pursuant to Rule 425 u	nder the Securities Act (17 CFR 230.425)		
☐ Soliciting material pursuant to Rule 14a-12 under	er the Exchange Act (17 CFR 240.14a-12)		
☐ Pre-commencement communications pursuant to	o Rule 14d-2(b) under the Exchange Act (17 C	FR 240.14d-2(b))	
☐ Pre-commencement communications pursuant to	Rule 13e-4(c) under the Exchange Act (17 Cl	FR 240.13e-4(c))	

TABLE OF CONTENTS

ITEM 1.01 — ENTRY INTO A MATERIAL DEFINITIVE AGREEMENT	
ITEM 7.01 — REGULATION FD DISCLOSURE	
ITEM 9.01 — FINANCIAL STATEMENTS AND EXHIBITS	4
SIGNATURES	;
EXHIBIT INDEX	(
2	

Item 1.01 Entry into a Material Definitive Agreement.

On May 16, 2016, ReWalk Robotics (the "Company" or "ReWalk") entered into a Research Collaboration Agreement (the "Collaboration Agreement") and an Exclusive License Agreement (the "License Agreement") with the President and Fellows of Harvard College ("Harvard").

Under the Collaboration Agreement, ReWalk and Harvard agree to collaborate on research regarding the development of lightweight "soft suit" exoskeleton system technologies for lower limb disabilities, which are intended to treat stroke, Multiple Sclerosis ("MS"), mobility limitations for the elderly and other medical applications. Under the Collaboration Agreement, the Company must pay Harvard quarterly installment payments to help fund the research. Subject to the terms of the Collaboration Agreement, the parties are required to report their respective research results and findings to each other on a regular basis. The Collaboration Agreement governs ownership of the research results and inventions generated in performance of the research collaboration, and provides the Company the option to negotiate with Harvard for a license to certain new inventions of Harvard conceived in performance of the collaboration.

The Collaboration Agreement will expire on May 16, 2021. Subject to payment of a minimum funding commitment under applicable circumstances, the Company may terminate the Agreement if there is a loss of Harvard's principal investigator or if the Company does not believe that it has or can secure sufficient funding to proceed. The Collaboration Agreement may also be terminated by either party due to a material uncured breach by the other party or upon termination of the License Agreement.

Under the License Agreement, ReWalk is granted an exclusive, worldwide royalty-bearing license under certain patents of Harvard relating to lightweight "soft suit" exoskeleton system technologies for lower limb disabilities, a royalty-free license under certain related know-how and the option to obtain a license under certain inventions conceived under the parties' joint research collaboration. Harvard retains the right to practice the patents for research, educational and scholarly purposes. The Company is required to use commercially reasonable efforts to develop products under the license in accordance with an agreed-upon development plan and to introduce and market such products commercially. In addition to an upfront fee and royalties on net sales, the Company is obligated to pay Harvard certain milestone payments upon the achievement of certain product development and commercialization milestones. The Company also agrees to reimburse Harvard for expenses incurred in connection with the filing, prosecution and maintenance of the licensed patents.

The License Agreement will continue in full force and effect until the expiration of the last-to-expire valid claim of the licensed patents. The Company may terminate the Agreement for any reason upon 60 days' prior written notice, while Harvard may terminate the Agreement if the Company does not obtain requisite insurance, becomes insolvent or fails to meet certain development milestones. The License Agreement may also be terminated by ReWalk or Harvard due to the other party's material uncured breach.

The Collaboration Agreement and License Agreement contain, as applicable, customary representations and warranties and customary enforcement, indemnification and insurance provisions.

These brief descriptions of the Collaboration Agreement and the License Agreement are qualified in their entirety by reference to the provisions of such agreements, which are attached to this current report on Form 8-K as Exhibits 10.1 and 10.2, respectively, and incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On May 17, 2016, the Company issued a press release titled "ReWalk Announces Collaboration with Harvard University's Wyss Institute to Develop Lightweight and Soft Exoskeleton Systems for the Treatment of Stroke, Multiple Sclerosis and Limited Mobility Patients," a copy of which is furnished as Exhibit 99.1 herewith and incorporated herein by reference.

The information in this Item 7.01, including Exhibit 99.1, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act") or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of the Company under the Securities Act of 1933, as amended, regardless of any general incorporation language in those filings.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits

$\mathbf{F}_{\mathbf{v}}$	hi	hit

Number Description

- Research Collaboration Agreement, dated May 16, 2016, between the Company and the President and Fellows of Harvard College. †
- License Agreement, dated May 16, 2016, between the Company and the President and Fellows of Harvard College.[†]
- 99.1 Press release titled "ReWalk Announces Collaboration with Harvard University's Wyss Institute to Develop Lightweight and Soft Exoskeleton Systems for the Treatment of Stroke, Multiple Sclerosis and Limited Mobility Patients," dated May 17, 2016.*

[†] Portions of this exhibit were omitted and a complete copy has been provided separately to the Securities and Exchange Commission pursuant to the Company's application requesting confidential treatment under Rule 24b-2 of the Exchange Act.

^{*} Furnished herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ReWalk Robotics Ltd.

By: /s/ Kevin Hershberger
Name: Kevin Hershberger

Title: Chief Financial Officer

Dated: May 17, 2016

EXHIBIT INDEX

Exhibit	
Number	Description
10.1	Research Collaboration Agreement, dated May 16, 2016, between the Company and the President and Fellows of Harv

- rvard College.[†]
- 10.2 License Agreement, dated May 16, 2016, between the Company and the President and Fellows of Harvard College.[†]
- Press release titled "ReWalk Announces Collaboration with Harvard University's Wyss Institute to Develop Lightweight and Soft Exoskeleton 99.1 Systems for the Treatment of Stroke, Multiple Sclerosis and Limited Mobility Patients," dated May 17, 2016.*
- [†] Portions of this exhibit were omitted and a complete copy has been provided separately to the Securities and Exchange Commission pursuant to the Company's application requesting confidential treatment under Rule 24b-2 of the Exchange Act.

* Furnished herewith.

RESEARCH COLLABORATION AGREEMENT

This Research Collaboration Agreement (together with all Exhibits hereto, this "**Agreement**"), effective as of May 16, 2016 (the "**Effective Date**"), is entered into by and between **President and Fellows of Harvard College**, a charitable corporation of Massachusetts having an office at Richard A. and Susan F. Smith Campus Center, Suite 727, 1350 Massachusetts Avenue, Cambridge, Massachusetts 02138 ("**Harvard**"), and **ReWalk Robotics**, **Ltd.** a company existing under the laws of the State of Israel, having a place of business at 200 Donald Lynch Blvd., Marlborough, MA 01752 ("**Company**"). Harvard and Company each shall be referred to herein as a "**Party**" and together as the "**Parties**."

WHEREAS, Harvard and Company desire to collaborate in the performance of research to be described in a research plan; and

WHEREAS, Harvard and Company believe that collaborating with each other in the performance of such research will be of mutual benefit, will further the instructional and research objectives of Harvard and will foster the development of scientific knowledge.

NOW, THEREFORE, in consideration of the promises and mutual covenants set forth herein, Harvard and Company agree as follows:

1. Definitions.

Whenever used in this Agreement with an initial capital letter, the terms defined in this <u>Article 1</u>, whether used in the singular or the plural, shall have the meanings specified below.

- **1.1** "Active" shall mean, with respect to a Soft Exosuit, technology that commands motors to apply torques at joint(s) beyond adjusting pretension levels.
- **1.2** "Calendar Quarter" shall mean each of the periods of three (3) consecutive calendar months ending on March 31, June 30, September 30 and December 31 during the Term.
- **1.3** "Company Invention" shall mean any patentable invention that is conceived and/or reduced to practice in the performance of the Research Plan during the Term for which at least one inventor is a member of the Company Team (but no inventor is a member of the Harvard Team).
- **1.4** "Company Patent Rights" shall mean any patents and patent applications that claim any Company Invention, in each case solely to the extent the claims are directed to the subject matter of such Company Invention.
- **1.5 "Company Principal Investigator"** shall mean Ofir Koren, or such other principal investigator who may replace him pursuant to Section 2.2.
 - **1.6** "Company Results" shall mean Results generated by the Company Team.
- **1.7** "**Company Team**" shall mean the Company Principal Investigator and those technicians, scientists and/or other individuals employed (or otherwise retained) by Company to work on behalf of Company under [his/her] direction on the Research.
- **1.8** "Control" shall mean, with respect to any Patent Rights, other intellectual property rights or trade secrets, the legal authority or right (whether by license or otherwise) of a Party to grant a license or a sublicense of or under such Patent Rights, or other intellectual property rights to another Person, or to otherwise disclose such trade secrets to another Person, without breaching the terms of any agreement with a third party, or misappropriating the trade secrets of a third party.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- 1.9 "Field" shall mean Active Medical Lower Limb Soft Exosuits.
- **1.10** "Governmental Authority" means any transnational, domestic or foreign federal, state or local governmental, regulatory or administrative authority, department, court, agency or official, including any political subdivision thereof.
- **1.11** "Harvard Background Patents" shall mean, in each case to the extent owned or Controlled by Harvard, all of the Patent Rights for the patents and patent applications listed in Exhibit C.
 - **1.12** "Harvard Inventions" shall mean Harvard Type A Inventions and Harvard Type B Inventions, collectively.
- **1.13** "Harvard Principal Investigator" shall mean Conor Walsh, Ph.D., or such other principal investigator who may replace Dr. Walsh pursuant to Section 2.2.
 - **1.14** "Harvard Results" shall mean Results generated by the Harvard Team.
- **1.15** "Harvard Team" shall mean the Harvard Principal Investigator and those faculty members, research fellows, students, technicians, scientists and/or other individuals employed (or otherwise retained) by Harvard to work on behalf of Harvard under his/her direction on the Research.
- **1.16** "Harvard Type A Invention" shall mean any patentable invention that either (a) is conceived and reduced to practice (i)(x) in the performance of the Research Plan during the Term or (y) conceived in the performance of the Research Plan during the Term and reduced to practice within one (1) year after termination or expiration of the Term, (ii) for which at least one inventor is a member of the Harvard Team (but no inventor is a member of the Company Team) and (iii) which is dominated by a Valid Claim within the Harvard Background Patents (regardless of whether or not such Valid Claim ultimately issues); or (b) is deemed a Harvard Type A Invention pursuant to Section 5.2.4.
- **1.17** "Harvard Type A Patent Rights" shall mean all of the Patents Rights for any patents or patent applications that claim any Harvard Type A Invention, in each case solely to the extent the claims are directed to the subject matter of such Harvard Type A Invention.
- **1.18** "Harvard Type B Invention" shall mean any patentable invention (i) conceived and reduced to practice during the Term or (ii) conceived in the performance of the Research Plan during the Term and reduced to practice within one (1) year after termination or expiration of the Term (a) that is not a Harvard Type A Invention, (b) for which an inventor is a member of the Harvard Team (but no inventor is a member of the Company Team). Harvard Type B Inventions shall exclude new actuator and sensor components.
- **1.19** "Harvard Type B Patent Rights" shall mean all of the Patent Rights for any patents or patent applications that claim any Harvard Type B Invention, in each case solely to the extent the claims are directed to the subject matter of such Harvard Type B Invention.
 - **1.20** "**Invention**" shall mean a Company Invention, Harvard Invention or Joint Invention, as the case may be.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.21** "Joint Invention" shall mean any patentable invention that is (i) conceived and reduced to practice in the performance of a Research Plan during the Term or (ii) conceived in the performance of the Research Plan during the Term and reduced to practice within [**] after termination or expiration of the Term, for which one or more inventors are members of the Company Team and one or more inventors are members of the Harvard Team.
- **1.22** "**Joint Patents**" shall mean any patents and patent applications that claim any Joint Invention, in each case solely to the extent the claims are directed to the subject matter of such Joint Invention.
 - **1.23** "License Agreement" shall mean the agreement entered into by the Parties as of the Effective Date.
- **1.24** "**Medical Lower Limb**" shall mean actively affecting leg movement in patients with gait impairments resulting from the lack of proper muscle function due to neurological impairment, muscular degeneration or advanced aging.
- **1.25** "**Person**" shall mean an individual, corporation, partnership, limited liability company, association, trust or other entity or organization, including a Governmental Authority.
 - **1.26** "Principal Investigators" shall mean the Company Principal Investigator and the Harvard Principal Investigator.
- 1.27 "Patent Rights" shall mean, with respect to any patents or patent applications, all of the following: (a) such patents and patent applications (including the PCT and/or U.S. utility application claiming priority to such application(s) that are filed on or before the one year conversion date of such application(s)); (b) any patent or patent application that claims priority to and is a divisional, continuation, reissue, renewal, reexamination, substitution or extension of any patent application identified in (a); (c) any patents issuing on any patent application identified in (a) or (b), including any reissues, renewals, reexaminations, substitutions or extensions thereof; (d) any claim of a continuation-in-part application or patent (including any reissues, renewals, reexaminations, substitutions or extensions thereof) that is entitled to the priority date of, and is directed specifically to subject matter specifically described in, at least one of the patents or patent applications identified in (a), (b) or (c); (e) any foreign counterpart (including PCTs) of any patent or patent application identified in (a), (b) or (c) or of the claims identified in (d); and (f) any supplementary protection certificates, pediatric exclusivity periods, any other patent term extensions and exclusivity periods and the like of any patents and patent applications identified in (a) through (e).
- **1.28** "Research" shall mean the research actually conducted during the Term by the Company Team and/or Harvard Team under the terms of this Agreement in accordance with the Research Plan.
- **1.29** "**Research Plan**" shall mean the research plan attached hereto as <u>Exhibit A</u> (as may be amended from time to time by mutual agreement of the Parties), which sets forth the scope, timing and details of the research to be performed by the Company Team and Harvard Team under the direction of their respective Principal Investigators during the Term.
- **1.30** "Results" shall mean all data, materials, compositions, methods, processes, analyses, formulae, know-how, trade secrets, unpatented inventions (whether patentable or unpatentable and whether or not reduced to practice), technology, and information generated in the performance of the Research, but excluding Inventions.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.31** "**Soft Exosuit**" shall mean assistive technology comprised of (a) a motor/gear-driven actuation system, (b) Bowden-cable-based transmission, (c) a textile-based human interface, (d) wearable sensors and (e) algorithms to control assistance.
- **1.32** "Valid Claim" shall mean: (a) a claim of an issued and unexpired patent that has not been (i) held permanently revoked, unenforceable, unpatentable or invalid by a decision of a Governmental Authority of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, (ii) rendered unenforceable through disclaimer or otherwise, (iii) abandoned or (iv) permanently lost through an interference or opposition proceeding without any right of appeal or review; or (b) a pending claim of a pending patent application that (i) has been asserted and continues to be prosecuted in good faith and (ii) has not been abandoned or finally rejected without the possibility of appeal or refiling.
- **1.33 Additional Defined Terms**. The following capitalized terms shall have the meanings ascribed to them in the following sections of this Agreement:

Abandoned Joint Patent	Section 6.2.5.2
Agreement	Preamble
Company	Preamble
Company Confidential Information	Section 7.1
Company Visiting Scientist	Section 2.3
Confidential Information	Section 7.1
Ex-Field Licensee	Section 10.16
Harvard	Preamble
Harvard Confidential Information	Section 7.1
Harvard Names	Section 10.5
Harvard Visiting Scientist	Section 2.3
MS	Exhibit A
Negotiation Period	Section 6.2.3
Option Activation Period	Section 6.2.2
Option Exercise Period	Section 6.2.3
Party or Parties	Preamble
Reimbursed Costs	Section 10.16
Suit	Section 10.10
Term	Section 9.1
Visiting Scientist	Section 2.3

2. Research.

- **2.1 Performance of Research**. The Parties shall use good faith efforts to perform the Research in accordance with the Research Plan; however, neither Party makes any warranties or representations regarding completion of the Research or the achievement of any particular results.
- **2.2 Principal Investigators**. The part of the Research to be performed by Harvard will be directed and supervised by the Harvard Principal Investigator, who shall have primary responsibility for the performance of such Research. The part of the Research to be performed by Company will be directed and supervised by the Company Principal Investigator, who shall have primary responsibility for the performance of such Research. The Principal Investigators shall maintain regular contact with each other, at least monthly, and shall facilitate the coordination of the Parties' activities under the Research Plan. Company may replace the named Company Principal Investigator upon prior written notice to Harvard. If the Harvard Principal Investigator ceases to supervise the Research for any reason, Harvard will notify Company promptly and may endeavor to find a substitute acceptable to Company. If Harvard declines or is unable to find a substitute acceptable to Company within [**] after the Harvard Principal Investigator ceases to supervise the Research, Company may terminate this Agreement immediately upon written notice to Harvard.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

2.3 Visiting Scientists. The Parties agree that one or more scientists from Company (each, a "Company Visiting Scientist") may perform or otherwise participate in Research at Harvard's facilities, and one or more scientists from Harvard (each, a "Harvard Visiting Scientist", and together with Company Visiting Scientists, the "Visiting Scientists") may perform or otherwise participate in Research at Company's facilities. Company shall provide Harvard with prior written notice of each Company Visiting Scientist that Company intends to have perform or otherwise participate in Research at Harvard's facilities. Each Visiting Scientist from Company shall execute a Visitor Participation Agreement, in the form attached hereto as Exhibit D), prior to his/her visit. Notwithstanding the terms of the Visitor Participation Agreement or any other provision of this Agreement to the contrary, the Parties hereby agree that if a Company Visiting Scientist conceives and reduces to practice any patentable invention in the performance of the Research Plan during the Term outside of Harvard's premises and facilities, then he/she shall assign his/her entire right, title and interest in and to such invention to Company. Company shall provide all necessary information and documentation required by the U.S. Department of Homeland Security, if any is required, and shall ensure that its Visiting Scientist has the appropriate visa for his/her visit(s) to Harvard. Each Party will endeavor to apprise the other Party's Visiting Scientists of any rules, regulations, procedures and policies that may be applicable to the Visiting Scientist's activities at its facilities. In any event, however, the Visiting Scientist shall comply with all applicable rules, regulations, procedures and policies at all times during the term of this Agreement. Neither Party shall compensate or reimburse the other Party or the other Party's Visiting Scientist in connection with the Visiting Scientist's visit(s) under this Agreement. Each Party shall pay and administer all compensation and fringe benefits, if any, due to its Visiting Scientists. Neither Party shall provide housing, meals, travel or any other personal expenses for the other Party's Visiting Scientists. Instead, all such expenses shall be the responsibility of the Visiting Scientist or his/her employer. Each Party shall provide health insurance for its Visiting Scientists. Each Company Visiting Scientist shall sign and deliver to Harvard, prior to the commencement of his/her visit, the "Acknowledgement of Risk and Release for Non-Harvard Personnel Using Harvard Research and Instructional Laboratory Facilities," in the form attached hereto as **Exhibit E**).

2.4 Harvard Team. Within [**] of the Effective Date, Harvard will identify in writing to Company all of the members of the Harvard Team. Harvard shall use reasonable efforts to ensure that the continued inclusion of such individuals on the Harvard Team. If any member of the Harvard Team will be removed or replaced, Harvard shall provide reasonable advance notice thereof to Company and shall provide Company with a qualified replacement member, having qualifications and experience at least equal to the former member, who is reasonably satisfactory to Company. If for any reason, Company objects to any new proposed member of the Harvard Team, the Parties shall work together in good faith to identify an alternative new member. At no point during the Term will there be less than [**] members of the Harvard Team.

3. Costs and Expenses.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- 3.1 Subject to Section 9.6.1, Company agrees to pay Harvard the total amount of [**] United States Dollars (US \$[**]), as follows:
- **3.1.1** [**] United States Dollars (US \$[**]) for 2016 to be paid in [**] equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within forty-five (45) days of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of each Calendar Quarter in 2016 after the Effective Date.
- **3.1.2** [**] United States Dollars for 2017 (\$[**]) to be paid as follows: (a) a first installment of [**] United States Dollars (US \$[**]) and (b) [**] subsequent equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within [**] of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of each Calendar Quarter in 2017.
- **3.1.3** [**] United States Dollars (\$[**]) for 2018 to be paid as follows: (a) a first installment of [**] United States Dollars (US \$[**]) and (b) [**] subsequent equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within [**] of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of each Calendar Quarter in 2018
- **3.1.4** [**] United States Dollars for 2019 (\$[**]) to be paid as follows: (a) a first installment of [**] United States Dollars (US \$[**]) and (b) [**] subsequent equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within [**] of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of each Calendar Quarter in 2019.
- **3.1.5** [**] United States Dollars (\$[**]) for 2020 to be paid as follows: (a) a first installment of [**] United States Dollars (US \$[**]) and (b) [**] subsequent equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within [**] of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of each Calendar Quarter in 2020.
- **3.1.6** [**] United States Dollars (\$[**]) for 2021 to be paid in [**] equal installments of [**] United States Dollars (US \$[**]) each, each of which shall be payable within [**] of receipt of the invoice from Harvard for such amount. Harvard shall issue each invoice to Company at least [**] prior to the start of the [**] Calendar Quarters of 2021.
- **3.2** Harvard shall use the amounts paid by Company to Harvard pursuant to this Article 3 solely for the conduct of the Research pursuant to the terms of this Agreement in accordance with the Research Plan. Harvard shall not be obligated to incur costs or expend funds to conduct the Research in excess of the total amount paid by Company under this Agreement.
- 3.3 Harvard will provide Company with a fixed-price invoice in the form set forth in Exhibit B hereto prior to each scheduled Calendar Quarterly payment; provided that Harvard's failure to do so will not relieve Company from its obligation to make the corresponding payment, but instead will entitle Company to delay the due date of such payment until the date that is [**] after the date of Harvard's invoice. All payments due under this Agreement will be paid in United States dollars and will be without deduction or withholding of any kind. To the extent that any terms or conditions of any invoice issued by Company in connection with Harvard's invoices under this Section 3.3 conflict with the terms and conditions of this Agreement, the terms and conditions of this Agreement will prevail and govern.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- 4. Results.
- 4.1 Reports; Use; License.
- **4.1.1** Each Party shall provide the other with reports summarizing its Results, no less frequently than once each Calendar Quarter. After each such exchange, the Parties shall meet either in person or by teleconference to review the Results. Within [**] after the earlier of the completion of the Research or the termination of this Agreement, each Party will provide the other with a final report summarizing its Results.
- **4.1.2** Subject to Articles 7 and 8 of this Agreement and without limiting any rights or licenses expressly granted by Harvard to Company pursuant to this Agreement or the License Agreement, Harvard shall have the right to use Company's Results for internal research purposes only.
- **4.1.3** Subject to Articles 7 and 8 of this Agreement and without limiting any rights or licenses expressly granted by Harvard to Company pursuant to this Agreement or the License Agreement, Harvard hereby grants to Company a perpetual, non-exclusive, royalty-free license to use any Harvard Results both (a) for internal research purposes and (b) to make, have made, offer for sale, sell, have sold, repair, service and import products and services within the Field in the Territory.
- **4.2 Confidentiality.** Without regard to the marking requirement described in Section 7.1 or the disclosure mechanism described in Section 7.2 and subject to the exceptions set forth in Section 7.1 (i) (iv), each Party (a) shall treat as Confidential Information (as otherwise defined in Section 7.1) of the other Party the contents of any report provided to it under Section 4.1 that discloses Results generated solely by the other Party and (b) shall treat as Confidential Information of the Parties jointly the contents of any report provided to it under Section 4.1 that discloses Results generated jointly by the Parties until publication of such joint Results in accordance with Article 8 (except that patent filings in accordance with Section 5.4 shall be permitted).
- **4.3 Meet and Confer.** If, during the Term, Company notifies Harvard Principal Investigator that Company believes that certain Harvard Results are competitively sensitive or would provide Company with a discernible competitive or market advantage that would be adversely affected if publicly used, disclosed or published, then the Parties shall use good faith efforts to meet and confer to discuss such Harvard Results and the concerns of Company regarding the use, disclosure and publication thereof by Harvard, the Harvard Principal Investigator or the Harvard Team.
 - 5. Inventions.
 - **5.1 Inventorship.** Inventorship of Inventions shall be determined in accordance with United States patent law.
 - 5.2 Ownership.
 - **5.2.1 Company Inventions**. The entire right, title and interest in and to all Company Inventions shall be owned solely by Company.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **5.2.2 Harvard Inventions**. The entire right, title and interest in and to all Harvard Inventions shall be owned solely by Harvard.
- **5.2.3 Joint Inventions**. The entire right, title and interest in and to all Joint Inventions shall be owned jointly by Company and Harvard. Subject to Section 6.2, Company and Harvard each shall have the full right in any jurisdiction to grant licenses under its interest in Joint Patents without any obligation to seek the consent of the other or to account for any profits made as a result of any such license.
- **5.2.4 Inventions Made by Company Visiting Scientists**. Notwithstanding anything to the contrary herein (but consistent with the fourth (4th) sentence of Section 2.3), patentable inventions conceived and reduced to practice by a Company Visiting Scientist solely during the period that such Company Visiting Scientist is performing or otherwise participating in Research at Harvard's facilities in the performance of the Research Plan will be deemed to be a Harvard Type A Invention.
- 5.2.5 **Inventions Made by Harvard Visiting Scientists.** Notwithstanding anything to the contrary herein, patentable inventions conceived and reduced to practice by a Harvard Visiting Scientist (other than Dr. Walsh) solely during the period that such Harvard Visiting Scientist is performing or otherwise participating in Research at Company's facilities in the performance of the Research Plan will be deemed to be a Joint Invention.
- **5.3 Disclosure**. Harvard shall notify Company, promptly and in writing, of any Invention with respect to which its Office of Technology Development has received a written invention disclosure form and (in the case of Harvard Inventions and Joint Inventions) filed a patent application; provided, however, that subject to and without limiting Sections 2.6.4 (with respect to Consulting Inventions and Joint Consulting Inventions (as such terms are defined in the License Agreement)) and 6.1 (with respect to any Inventions included in the Licensed Patent Rights (as such term is defined in the License Agreement)) of the License Agreement, Harvard may elect to notify Company prior to filing. Company shall notify Harvard, promptly and in writing, of any Company Invention for which it has filed a patent application; provided, however, that Company may elect to notify Harvard prior to filing. Without regard to the marking requirement described in Section 7.1 or the disclosure mechanism described in Section 7.2, and subject to the exceptions set forth in Section 7.1 (i) (iv), each Party (a) shall treat as Confidential Information of the other Party the contents of any notice provided to it under this Section 5.3 that discloses an Invention owned solely by the other Party and (b) shall treat as Confidential Information of the Parties jointly the contents of any notice provided to it under this Section 5.3 that discloses a Joint Invention until publication of such Joint Invention in accordance with Article 8 (except that patent filings in accordance with Section 5.4 shall be permitted).

5.4 Patent Filing and Prosecution.

5.4.1 Sole Inventions. Company shall be responsible, at its sole expense and discretion, for the preparation, filing, prosecution and maintenance of Company Patent Rights. Subject to and without limiting Section 6 of the License Agreement with respect to any Inventions included in the Licensed Patent Rights (as such term is defined in the License Agreement), Harvard shall be responsible, at its sole expense and discretion, for the preparation, filing, prosecution and maintenance of Harvard Patent Rights.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **5.4.2 Joint Inventions.** Harvard shall be responsible for the preparation, filing, prosecution and maintenance of Joint Patents; provided, however, that Harvard may elect to waive such right on a case-by-case basis, in which case Harvard shall notify Company promptly in writing and Company shall have the right, but not the obligation, to take responsibility for such Joint Patents at its own expense. The responsible Party shall: (a) use outside patent counsel reasonably acceptable to the other Party and instruct such patent counsel to furnish the other Party with copies of all correspondence relating to Joint Patents from all patent offices, as well as copies of all proposed responses to such correspondence in time for the other Party to review and comment on such responses; (b) give the other Party an opportunity to review the text of each patent application before filing; (c) consult with the other Party with respect thereto; (d) supply the other Party with a copy of the application as filed, together with notice of its filing date and serial number; and (e) keep the other Party advised of the status of actual and prospective patent filings. The responsible Party shall give the other Party the opportunity to provide comments on and make requests of the responsible Party concerning the preparation, filing, prosecution and maintenance of Joint Patents, and shall consider such comments and requests in good faith.
- **5.5 No License or Grant of Rights**. Except as expressly provided in <u>Section 4.1</u> or <u>Article 6</u>, nothing in this Agreement shall be construed to confer any ownership interest, license or other rights upon a Party by implication, estoppel or otherwise as to any technology, intellectual property rights, products or biological materials of the other Party or any other entity.
 - 6. Intellectual Property.
- **6.1 Harvard Type A Inventions**. Harvard shall grant a license to Company under Harvard's interest under the Harvard Type A Inventions pursuant to and in accordance with the License Agreement.

6.2 Grant of Option Under Harvard Type B Inventions and Harvard's Interest in Joint Patents.

- **6.2.1 Option**. With respect to each Harvard Type B Invention, Harvard shall notify Company of such Harvard Type B Invention prior to notifying any third party and Harvard hereby grants to Company the first option to negotiate in good faith with Harvard for a non-exclusive or an exclusive (at Company's discretion), royalty-bearing, worldwide license under the Harvard Type B Patent Rights to develop, make, have made, offer for sale, sell, have sold and import products and services in the Field on terms that are commercially reasonable to the industry; provided, however, that no such license will include any grant of exclusive rights that would violate the National Institutes of Health's Principles and Guidelines for Recipients of NIH Research Grants and Contracts on Obtaining and Disseminating Biomedical Research Resources, as published at 64 Fed. Reg. 72090, and as may be amended from time to time. With respect to each Joint Invention, Harvard shall notify Company of its intent to file any application for a Joint Patent and Harvard hereby grants to Company the first option to negotiate in good faith with Harvard for an exclusive license under Harvard's interest in the Joint Patent to develop, make, have made, offer for sale, sell, have sold and import products and services in the Field on terms that are commercially reasonable to the industry.
- 6.2.2 Activation; Patent Costs. If Company chooses to activate such option to negotiate with Harvard for a license under a Harvard Type B Invention (or, if applicable, an exclusive license under Harvard's interest in a Joint Invention), it shall provide Harvard with written notice thereof within [**] after disclosure of the relevant Harvard Type B Invention (or relevant Joint Invention) under Section 5.3 (the "Option Activation Period"). If Company activates an option during the relevant Option Activation Period, then subject to Section 10.16, it shall reimburse Harvard for all reasonable documented, out-of-pocket costs relating to the preparation, filing, prosecution and maintenance of the relevant Harvard Type B Patent Rights (or relevant Joint Patent) incurred by Harvard prior to activation of the option, as well as those incurred during the Option Exercise Period and Negotiation Period. Harvard shall submit periodic invoices and Company shall make payment within [**] after the date of each such invoice.

q

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- 6.2.3 Exercise. If Company chooses to exercise its option to negotiate with Harvard for a license under a Harvard Type B Invention (or, if applicable, an exclusive license under Harvard's interest in a Joint Invention) that Company has activated under Section 6.2.2, Company must provide Harvard with written notice of such decision at any time during the Term or within [**] after expiration or termination of the Term (collectively, the "Option Exercise Period,"). If Company exercises an option to negotiate with Harvard for a license under a Harvard Type B Invention within the Option Exercise Period, it shall have [**] after such exercise within which to execute a license agreement with Harvard (the "Negotiation Period"); provided that the Negotiation Period may be extended by mutual agreement of the Parties. Each of Harvard and Company shall use good faith efforts to negotiate and execute such mutually-agreeable license agreement. If Company exercises an option for an exclusive license under Harvard's interest in a Joint Invention, then upon receipt by Harvard of the written request of Company, the Parties shall immediately amend Exhibit 1.25 of the License Agreement to add Harvard's interest in such Joint Patent and upon such amendment, each such Joint Patent shall automatically (and without any action of the Parties) be deemed to be "Licensed Patent Rights" licensed pursuant to the License Agreement by Harvard for all documented, out-of-pocket expenses incurred by Harvard in connection with such Licensed Patent Rights in accordance with Article 6 thereof.
- 6.2.4 Terms. Except as otherwise expressly agreed by the Parties in writing, such license agreement for a license under a Harvard Type B Invention shall include, without limitation, (a) in the case of an exclusive license, terms consistent with the provisions of 35 U.S.C. §§ 200-212 and 37 C.F.R. § 401 et seq., and a reservation of the rights of Harvard and other not-for-profit research organizations to practice the subject matter of the licensed Harvard Type B Patent Rights for research, teaching and other educational purposes only, (b) the indemnity, insurance, limitations on liability, patent cost reimbursement (subject to a provision consistent with Section 10.16) and other provisions customary to patent and technology licenses normally granted by Harvard (i.e., consistent with those as set forth in the License Agreement), and (c) commercially reasonable due diligence obligations for the development and commercialization of products or processes covered by the relevant Harvard Type B Patent Rights (i.e., consistent with those set forth in the License Agreement).

6.2.5 Expiration.

6.2.5.1 With respect to each Harvard Type B Invention, if Company (a) does not activate its option during the relevant Option Activation Period, (b) provides written notice to Harvard during the relevant Option Activation Period that it waives its option in whole or in part, (c) activates its option during the relevant Option Activation Period but does not exercise the Option during the Option Exercise Period or (d) exercises its Option during the Option Exercise Period and the Parties fail to reach agreement on terms and conditions of a license agreement within the Negotiation Period, then Company's rights under this Article 6 with respect to such Harvard Type B Invention shall expire.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

6.2.5.2 With respect to each Joint Invention, if Company decides that it does not wish to pay for the preparation, filing, prosecution or maintenance of a Joint Patent in a particular country (each, an "**Abandoned Joint Patent**"), Company shall provide Harvard with prompt written notice of such election. Upon Harvard's receipt of such notice, Company shall be released from its obligation to reimburse Harvard for the expenses incurred thereafter as to such Abandoned Joint Patents; provided, however, that expenses authorized prior to the receipt by Harvard of such notice shall be deemed incurred prior to the notice. For clarity, Company's right thereafter under such Abandoned Joint Patents shall be limited to a non-exclusive license in the Field in the relevant territory solely to the extent necessary for Company to practice other Patent Rights dominated by a Valid Claim within such Abandoned Joint Patent.

7. Confidential Information.

- **7.1 Definitions.** "**Confidential Information**" shall mean all information (including Results, but only until publication in accordance with <u>Article 8</u> hereof) that is marked as confidential (or, if disclosed orally or in intangible form, that is summarized in a writing that is marked as confidential and delivered to the recipient within thirty (30) days after disclosure) and that is disclosed (a) by or on behalf of Harvard (including by any member of the Harvard Team) to Company hereunder ("**Harvard Confidential Information**") or (b) by or on behalf of Company (including by any member of the Company Team) to Harvard or the Harvard Principal Investigator hereunder ("**Company Confidential Information**"). Notwithstanding the above, the obligations set forth in <u>Sections 7.2</u>, <u>7.3</u> and <u>8.2</u> shall not apply to Confidential Information to the extent that it: (i) was known to the recipient at the time it was disclosed, other than by previous disclosure by or on behalf of the discloser, as evidenced by written records at the time of disclosure; (ii) is at the time of disclosure or later becomes publicly known under circumstances involving no breach of this Agreement; (iii) is lawfully and in good faith made available to the recipient by a third party who is not subject to obligations of confidentiality to the discloser with respect to such information; or (iv) is independently developed by the recipient without the use of or reference to Confidential Information, as demonstrated by documentary evidence.
- **7.2 Disclosure Limitation.** Pursuant to Harvard policy, the Harvard Principal Investigator is not supposed to receive information that is subject to confidentiality obligations, including any Company Confidential Information, if doing so would affect his ability to publish research results or the ability of other scholars to replicate the published results. Accordingly, Company agrees to disclose Company Confidential Information to the Harvard Principal Investigator only if (a) it first notifies the Harvard Principal Investigator of the nature of such information and (b) the Harvard Principal Investigator, in his sole discretion, notifies Company that he wishes to accept the specified information, or a portion thereof. For clarity the obligations of the Harvard Principal Investigator set forth in Sections 7.3 and 8.2 shall only apply to Company Confidential Information accepted by the Harvard Principal Investigator in accordance with this Section 7.2.
- 7.3 **Obligations.** Company, Harvard, and the Harvard Principal Investigator each agree that, without the prior written consent of Harvard (in the case of Company being the recipient) or Company (in the case of Harvard or the Harvard Principal Investigator being the recipient) in each case, during the Term of this Agreement, and for five (5) years thereafter, it/he (a) will not disclose Confidential Information that it/he has received hereunder to any third party and (b) will not use Confidential Information that it/he has received hereunder except for the purposes of performing the Research and, in the case of Company, evaluating whether to exercise an option under Article 6 of this Agreement. Company, Harvard, and the Harvard Principal Investigator each shall treat Confidential Information that it/he has received hereunder with the same degree of confidentiality as it/he treats its/his own confidential and proprietary information, but in all events no less than a reasonable degree of confidentiality. Company, Harvard and the Harvard Principal Investigator each may disclose Confidential Information that it/he has received hereunder only to members of the Company Team or Harvard Team, respectively, who have a need to know such information for the purposes specified above and who agree in writing to protect such Confidential Information in accordance with the terms set forth in this Agreement.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

7.4 Agreement. The terms (but not the existence) of this Agreement constitute the Confidential Information of both Parties and neither Party shall disclose the terms of this Agreement to any third party except (a) to its Affiliates, Distributors and Sublicensees (as such terms are defined in the License Agreement) and their respective officers, directors, members, employees, agents and outside advisors who reasonably need to know such information for exercising such Party's rights or performing such Party's obligations under this Agreement or the License Agreement, (b) with the prior written approval of the other Party, (c) the Parties may issue a press release substantially in the form set forth in Exhibit F, which press release has been approved in advance by Harvard, or (d) this restriction shall not apply to any information required by law to be disclosed to any Governmental Authority or required to be disclosed publicly pursuant to applicable law, including, without limitation, pursuant to or in connection with securities law reporting obligations or the rules of any securities exchange, provided that the Party disclosing such information shall seek confidential treatment of such Confidential Information to the extent permitted under applicable law. Notwithstanding the foregoing, Harvard may publicly acknowledge Company's support for the investigations being pursued under this Agreement and in such statement, may disclose a high-level description of the research project (without disclosing specific timelines or deadlines), the relevant Harvard school/researchers, and the identity of Company as the sponsor).

8. Publications.

- **8.1** It is possible that Harvard and Company will publish the Results jointly. Nonetheless, subject to the provisions of this Agreement (including <u>Article 7</u>), each Party reserves the right to publish its Results separately. Harvard shall provide Company with a copy of any manuscript or other publication disclosing Results at least [**] prior to submission for publication for the purpose of enabling Company to review the manuscript or other publication for potentially patentable Inventions with respect to which it wishes to exercise its option rights under <u>Article 6</u> and/or for Company Confidential Information. Company shall provide Harvard with a copy of any manuscript or other publication disclosing Results at least [**] prior to submission for publication for the purpose of enabling Harvard to review the manuscript or other publication for potentially patentable Joint Inventions with respect to which it would like to file a patent application and/or for Harvard Confidential Information.
- **8.2** The Party wishing to publish shall delete from its manuscript or other publication prior to submission all Confidential Information of the other Party that the other Party identifies and requests the Party wishing to publish to delete within the [**] period specified in <u>Section 8.1</u>.
- **8.3** If, during the [**] period specified in Section 8.1, Company notifies Harvard that a manuscript or other publication reveals a potentially patentable Harvard Type B Invention or Joint Invention for which it wishes to exercise an option pursuant to Article 6, or Harvard notifies Company that a manuscript or other publication reveals a potentially patentable Joint Invention with respect to which it would like to file a patent application, the Party wishing to publish shall delay publication for the purpose of enabling a patent application to be filed in accordance with Section 5.4 until the earliest to occur of: (a) a patent application has been filed with respect to such Harvard Type B Invention or Joint Invention; (b) Harvard's Office of Technology Development and Company have determined that the relevant Harvard Type B Invention or Joint Invention is not patentable; or (c) [**] have elapsed from the date of notification under this Section 8.3.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

8.4 Notwithstanding anything to the contrary herein, the Parties agree to abide by the policies of journals in which publications will appear as to such matters as the public release or availability of data or biological materials relating to the manuscript or other publication. Proper acknowledgment will be made for the contributions of each Party to the Results being published. In addition, to the extent required by applicable journal policies, each Party shall use reasonable efforts to make samples of its research materials disclosed in the manuscript or other publication available upon request (supplies permitting) to scientists at non-profit institutions, provided that the recipient scientist agrees in writing that such research materials (a) will be used for research in the recipient scientist's laboratory only, (b) will not be used for any commercial purpose, (c) will not be used for work on human subjects, and (d) will not be distributed to other laboratories.

9. Term and Termination.

9.1 Term. This Agreement shall commence on the Effective Date and shall remain in effect for a period of five (5) years (the "**Term**"), unless earlier terminated in accordance with the provisions of this <u>Article 9</u>.

9.2 Termination.

- **9.2.1 Loss of Harvard Principal Investigator.** In the event that Dr. Walsh ceases to supervise the Research and Harvard declines or is unable to find a substitute the Harvard Principal Investigator acceptable to Company as provided in <u>Section 2.2</u>, Company may terminate this Agreement in accordance with <u>Section 2.2</u>.
- **9.2.2 Termination for Default.** In the event that either Party commits a material breach of its obligations under this Agreement and fails to cure that breach within thirty (30) days after receiving a written demand to cure from the non-breaching Party, the non-breaching Party may terminate this Agreement immediately upon written notice of termination to the breaching Party.
- **9.2.3 Termination for Convenience.** Subject to provisions of <u>Section 9.6.1</u> regarding the payment by Company of any minimum commitment, Company may terminate this Agreement if the Company does not believe that it either has sufficient finances to proceed or can otherwise secure sufficient funding to proceed, upon sixty (60) days prior written notice to Harvard.
- **9.3 Cross-Termination.** This Agreement shall automatically (and without any action of the Parties) terminate upon termination of the License Agreement in accordance with Section 9.2 thereof.
- **9.4 Force Majeure.** Neither Party will be responsible for delays resulting from causes beyond its reasonable control, including, without limitation, fire, explosion, flood, war, strike or riot; provided that the non-performing Party uses commercially reasonable efforts to avoid or remove such causes of non-performance and continues performance under this Agreement with reasonable dispatch whenever such causes are removed.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **9.5 Survival.** The following provisions, as well as any rights, obligations and duties which by their nature extend beyond the expiration or termination of this Agreement, shall survive the expiration or termination of this Agreement: Article 7 and 9 and Sections 4.1.2, 4.2, 5.2, 5.3, 10.2, 10.3, 10.4, and 10.5. In addition, the provisions of Article 6 shall survive termination of this Agreement as necessary to effectuate the rights of Company, unless Harvard has terminated this Agreement because of a material uncured breach by Company. In addition, Sections 4.1.3 and 10.6 shall survive the expiration or termination of this Agreement but solely until expiration or termination of the License Agreement.
- **9.6 Effect of Termination.** Except as otherwise expressly set forth in <u>Section 9.5</u> or this <u>Section 9.6</u>, all rights and obligations of the Parties shall cease immediately upon expiration or termination of this Agreement.

9.6.1 Payment Obligations.

- 9.6.1.1 Payment of Minimum Commitment for Certain Terminations Prior to April 1, 2018. In the event that (a) Harvard terminates this Agreement in accordance with Section 9.2.2 for the uncured material breach of Company, or (b) Company terminates this Agreement in accordance with Section 9.2.3, where, in either case of (a) or (b), the effective date of such termination is prior to [**] then notwithstanding that such termination may predate the dates for invoicing and payment set forth in Section 3.1, Company's obligation to pay Harvard all amounts due pursuant to Sections 3.1.1, 3.1.2 and 3.1.3(a) hereof (which, for clarity, equals [**] U.S. Dollars (\$[**])) shall expressly survive such termination and Company shall pay Harvard all such amounts due no later than the respective deadlines set forth therein. For clarity, if this Agreement is terminated for any reason specified in this Section 9.6.1.1, Company shall have no obligation to pay any other amounts due or payable under Section 3.1 from and after the effective date of such termination.
- **9.6.1.2 Payment Obligations with respect to All Other Terminations.** For clarity, in the event that this Agreement is terminated for any other reason (including, (a) the Company terminates this Agreement in accordance with (i) Section 9.2.1 for loss of Harvard Principal Investigator, or (ii) Section 9.2.2 for the uncured material breach of Harvard, (b) the Company terminates this Agreement in accordance with Section 9.2.3 (where the effective date of such termination for convenience is on or after [**]), (c) Harvard terminates this Agreement in accordance with Section 9.2.2 for the uncured material breach of Company (where the effective date of such termination is on or after [**]), or (d) this Agreement automatically terminates in accordance with Section 9.3 due to termination of the License Agreement), then, in each such case, Company shall have no obligation to pay any amounts due or payable under Section 3.1 from and after the effective date of such termination. Notwithstanding anything to the contrary herein, in the event that this Agreement is terminated for the reasons specified in subparts (b) or (c) of this Section 9.6.1.2, Company shall pay Harvard the amount of any financial commitments reasonably incurred by Harvard prior to termination in connection with the Research that cannot be canceled, including, without limitation, graduate student and post-doctoral stipends provided that (x) Harvard shall use reasonable efforts not to incur any financial commitments that cannot be canceled with thirty (30) (or more) days prior written notice and (y) Harvard shall notify Company of any such material financial commitments periodically during the Term and upon Company's reasonable request.
 - **9.6.2 Title to Equipment and Materials.** Upon termination or expiration of this Agreement for any reason, Harvard shall retain title to all equipment or material purchased or fabricated by Harvard, the Principal Investigator or the Harvard Team (regardless of whether such equipment or materials were purchase or fabricated with funds provided by Company hereunder).

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

10. Miscellaneous.

- 10.1 Representations and Warranties.
 - **10.1.1 Mutual Representations.** Each Party represents and warrants to the other Party as follows:
- **(a) Duly Organized.** Such Party is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation, is qualified to do business and is in good standing as a foreign corporation in each jurisdiction in which the conduct of its business or the ownership of its properties requires such qualification and failure to have such would prevent such Party from performing its obligations under this Agreement.
- **(b) Due Authorization; Binding Agreement.** The execution, delivery and performance of this Agreement by such Party have been duly authorized by all necessary corporate action. This Agreement is a legal and valid obligation binding on such Party and enforceable in accordance with its terms and does not: (i) to such Party's knowledge and belief, violate any law, rule, regulation, order, writ, judgment, decree, determination or award of any court, governmental body or administrative or other agency having jurisdiction over such Party; nor (ii) conflict with, or constitute a default under, any agreement, instrument or understanding, oral or written, to which such Party is a party or by which it is bound.
- **(c) Employee/Contractor Agreements**. All of such Party's and its employees acting on its behalf pursuant to this Agreement are and will be obligated under a binding written agreement to assign to such Party or its designee all inventions and to comply with obligations of confidentiality and non-use consistent in scope with those set forth herein.
 - **10.1.2 Representations by Harvard.** OTD, on behalf of Harvard, represents, warrants and covenants to Company that as follows:
- (a) Third Party Rights. As of the Effective Date, to the knowledge of OTD, Harvard has not received any funding, participation or other rights from any third Person or other Governmental Authority that conflicts or is inconsistent with the rights, licenses and options granted to Company hereunder. As of the Effective Date, Harvard has the right to grant the licenses and rights as contemplated under this Agreement and has not granted any right to any third Person which would conflict or be inconsistent with the licenses and rights granted to Company hereunder. Harvard will not during the Term grant any right to any third Person which would conflict or be inconsistent with the licenses and rights granted to Company hereunder.
- **(b) Patents.** Harvard is the sole and exclusive owner of all right, title and interest in and to the Harvard Background Patents, including all of the patents and patent applications identified as "Harvard Background Patents" in Exhibit 1.25. All official fees, maintenance fees and annuities for the Licensed Patent Rights have been paid through the Effective Date.
- (c) Non-Infringement of Third Party Rights. OTD has not received any written notice from any Person of any actual or threatened claim that the use or practice of the Licensed Patent Rights or Licensed Know-How infringes or otherwise violates the intellectual property rights of a third Person.

1.5

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **(d) Patent and Technology Status.** As of the Effective Date, none of the Licensed Patent Rights is currently involved in any interference, reissue, reexamination, or opposition proceeding and OTD has not received any written notice from any Person of such actual or threatened proceeding.
- **(e) Non-Infringement by Third Parties**. As of the Effective Date, to the knowledge of OTD, no third Persons are infringing the Licensed Patent Rights or Licensed Know-How.
- WARRANTIES, EXPRESS OR IMPLIED, AS TO ANY MATTER RELATING TO THIS AGREEMENT, INCLUDING, WITHOUT LIMITATION: RESULTS; THE PERFORMANCE, CONDITION, ORIGINALITY OR ACCURACY OF THE RESEARCH OR MATERIALS; THE AVAILABILITY OF LEGAL PROTECTION FOR INVENTIONS OR ANY OTHER WORK PRODUCT OF THE RESEARCH; OR THE VALIDITY OR ENFORCEABILITY OF ANY PATENT RIGHTS. EXCEPT AS EXPRESSLY SET FORTH IN THIS ARTICLE 10, ALL MATERIALS PROVIDED HEREUNDER ARE PROVIDED "AS IS," AND NEITHER PARTY MAKES ANY WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE FOR ANY RESULTS OR MATERIALS PROVIDED HEREUNDER, OR THAT THE USE OF THE RESULTS OR MATERIALS WILL NOT INFRINGE ANY PATENT RIGHTS OR OTHER INTELLECTUAL PROPERTY RIGHT
- Responsibilities and Indemnification. Each Party shall be responsible for its own acts in the performance of the Research, its use of Results, and its use, storage and disposal of any materials. Company shall indemnify, defend and hold harmless Harvard and its current and former directors, governing board members, trustees, officers, faculty, medical and professional staff, employees, students, and agents and their respective successors, heirs and assigns from and against any third party claim, liability, cost, expense, damage, deficiency, loss or obligation of any kind or nature (including, without limitation, reasonable attorneys' fees and other costs and expenses of litigation) based upon, arising out of, or otherwise relating to Company's use of Results, including without limitation any cause of action relating to product liability, except to the extent caused by or resulting from the gross negligence or willful misconduct of Harvard.
- **10.4 Limitation of Liability.** Except with respect to Company's indemnification obligations under Section 10.3, neither Party will be liable to the other with respect to any subject matter of this Agreement under any contract, negligence, strict liability or other legal or equitable theory for (a) any indirect, incidental, consequential or punitive damages or lost profits or (b) cost of procurement of substitute goods, technology or services. Except with respect to Company's indemnification obligations under Section 10.3, each Party's aggregate liability for all damages of any kind arising out of or relating to this Agreement or its subject matter shall not exceed the amounts paid by Company to Harvard under this Agreement.
- **10.5 Use of Names**. Except as provided below, Company shall not, and shall ensure that its affiliates shall not, use or register the name "Harvard" or "Wyss Institute" (alone or as part of another name) or any logos, seals, insignia or other words, names, symbols or other indicia of origin owned by Harvard that, in each case, identify Harvard or the Wyss Institute or any other Harvard school, unit, division or affiliate (collectively, "**Harvard Names**") for any purpose as a trademark except with the prior written approval of, and in accordance with restrictions required by, Harvard, provided, however that Company may issue a press release substantially in the form set forth in Exhibit F, which press release has been approved in advance by Harvard. This restriction shall not apply to any information required by law to be disclosed to any Governmental Authority, including, without limitation, information required to be disclosed pursuant to or in connection with securities reporting. If, notwithstanding this prohibition, Company or any of its affiliates registers any Harvard Name as a trademark, service mark, domain name, trade name, business or company name or otherwise anywhere in the world, then, in addition to any other remedies Harvard may have, Harvard shall have the right to compel Company or such affiliate to assign its rights in such registration to Harvard and Company shall take such steps as may be necessary to transfer record ownership of such registration to Harvard, at Company's cost.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- Research Partially Funded by Grants. To the extent that any Invention has been partially funded by the federal government (including any of its agencies), this Agreement and the grant of any rights in such Invention is subject to and governed by federal law, such as the provisions of 35 U.S.C. §§ 200-212 and all associated implementing regulations, as well as all applicable terms and conditions of such grant, provided, however that neither Harvard, the Harvard Principal Investigator nor any member of the Harvard Team shall accept any funding that would conflict or otherwise be inconsistent with the rights and licenses granted to Company hereunder or in the License Agreement.
- **10.7 Independent Contractors**. Company will not have the right to direct or control the activities of Harvard or the Harvard Principal Investigator in performing the Research. Harvard will not have the right to direct or control the activities of Company or the Company Principal Investigator in performing the Research. Harvard and Company shall act hereunder only as independent contractors, and nothing herein contained shall give either Party the right to bind the other, be deemed to constitute either Party as agent for or partner of the other or any third party, or be construed or fiduciary relationship between the Parties.
- **10.8 Notices.** Any notices to be given hereunder shall be sufficient if signed by the Party giving same and delivered in one of the following manners: (a) hand delivery; (b) certified mail, return receipt requested; (c) expedited delivery via a nationally recognized courier service; (d) facsimile if the sender retains evidence of successful transmission and if the sender promptly sends the original by ordinary mail, or (e) email, if the sender retains evidence of the sent transmission and if the sender promptly sends the original by ordinary mail, in any event to the following addresses:

If to Licensee: ReWalk Robotics, Ltd.

200 Donald Lynch Blvd.

Marlborough, Massachusetts 01752

Email: <u>larry.jasinski@rewalk.com</u> and <u>ofir@rewalk.com</u>

Facsimile: 508 251-2970

Attn.: Larry Jasinski and Ofir Koren

With copy to the following (if a ReWalk Robotics, Ltd.

technical notice):

Hatnufa st. 3, floor 6

Yokneaam, POB 161, 2069203

Israel

Facsimile: +972-4-959 0125

Email: larry.jasinski@rewalk.com and ofir@rewalk.com

Attn.: Larry Jasinski and Ofir Koren

If to Harvard: Office of Technology Development

Harvard University

Richard A. and Susan F. Smith Campus Center, Suite 727

1350 Massachusetts Avenue Cambridge, Massachusetts 02138 Facsimile: (617) 495-9568

Attn.: Chief Technology Development Officer

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

By such notice, either Party may change its address for future notices. Notices mailed shall be deemed given on the date postmarked on the envelope. Notices sent by expedited delivery shall be deemed given on the date received by the courier, as indicated on the shipping manifest or waybill. Notices sent by fax shall be deemed given on the date faxed.

- **10.9 Modification.** No modification or waiver of this Agreement or of any covenant, condition or limitation herein contained shall be valid unless in writing and executed by duly-authorized representatives of both Parties. A failure by a Party to assert its rights under, including upon any breach or default of, this Agreement shall not be deemed a waiver of such rights. No such failure or waiver in writing by either Party with respect to any rights shall extend to or affect any subsequent breach or impair any right consequent thereon.
- **10.10 Governing Law and Venue.** This Agreement will be governed by, and construed in accordance with, the substantive laws of the Commonwealth of Massachusetts, without giving effect to any choice or conflict of law provision, except that questions affecting the construction and effect of any patent shall be determined by the law of the country in which the patent shall have been granted. Any action, suit or other proceeding arising out of or relating to this Agreement, its subject matter, any document or instrument delivered pursuant to, or a breach of this Agreement or any such document or instrument (a "Suit") shall be brought in a court of competent jurisdiction in the Commonwealth of Massachusetts, and the Parties hereby consent to the sole jurisdiction of the state and federal courts sitting in the Commonwealth of Massachusetts. Each Party agrees not to raise any objection at any time to the laying or maintaining of the venue of any Suit in any of the specified courts, irrevocably waives any claim that Suit has been brought in any inconvenient forum and further irrevocably waives the right to object, with respect to any Suit, that such court does not have any jurisdiction over such Party.
- **10.11 Severability.** If any provision of this Agreement is or becomes invalid or is ruled invalid by any court of competent jurisdiction or is deemed unenforceable, such invalidity shall not render the entire Agreement unenforceable or invalid but rather the Agreement shall be read and construed as if the invalid or unenforceable provision(s) are not contained herein, and the rights and obligations of the Parties shall be construed and enforced accordingly.
- **10.12** Assignment. This Agreement shall be binding upon and inure to the benefit of the respective successors and assigns of the Parties hereto; provided, however that neither Party may assign any of its rights or obligations under this Agreement to any other person or entity without the prior written consent of the other except that Company may, without such consent, assign this Agreement, in its entirety, and its rights, obligations and interests to (a) to any affiliate or (b) to any purchaser of all of Company's equity or all or substantially all of Company's assets or business to which this Agreement relates, or (c) to any successor corporation resulting from any merger or consolidation of Company with or into such corporation, provided, in each case, that the assignee agrees in writing to be bound by the terms of this Agreement. Any assignment purported or attempted to be made in violation of the terms of this Section 10.12 shall be null and void and of no legal effect.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **10.13 Entire Agreement.** This Agreement is the sole agreement with respect to the subject matter hereof and supersedes all other agreements and understandings between the Parties with respect to the same.
- **10.14 Counterparts.** The Parties may execute this Agreement in two or more counterparts, each of which shall be deemed an original, but both of which together shall constitute one and the same instrument. Transmission by facsimile or electronic mail of an executed counterpart of this Agreement shall be deemed to constitute due and sufficient delivery of such counterpart. If by electronic mail, the executed Agreement must be delivered in a .pdf format.
- **10.15 No Third-Party Beneficiaries.** This Agreement is for the sole benefit of the Parties hereto and their respective successors and permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other individual, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association or other entity any legal or equitable right, benefit or remedy of any nature whatsoever, under or by reason of this Agreement.
- **10.16 Reimbursement.** To the extent that Company reimburses Harvard for any costs incurred by Harvard for the preparation, filing, prosecution and maintenance of any Patent Rights hereunder (such costs, "**Reimbursed Costs**"), then if Harvard grants a license under such Patent Rights to a third Person outside the Field (each, an "**Ex-Field Licensee**") and receives reimbursement from such Ex-Field Licensee for Reimbursed Costs, then Company shall only be responsible for reimbursing an appropriate portion of such Reimbursed Costs based on the scope and type of license (i.e., non-exclusive vs. exclusive) granted to such Ex-Field Licensee and Harvard shall credit Company's portion of the Reimbursed Costs to amounts owed under the License Agreement.

[Signature page follows]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

IN WITNESS WHEREOF, each Party has caused this Agreement to be executed by its duly authorized representative as of the Effective Date.

President and Fellows of Harvard College		ReWalk Robotics, Ltd.	
By:	/s/ Isaac T. Kohlberg	By:	/s/ Larry Jasinski
Name:	Isaac T. Kohlberg	Name:	Larry Jasinski
	Senior Associate Provost		
	Chief Technology Development Officer		
	Office of Technology Development		
Title:	Harvard University	Title:	CEO
including /s/ Conor		ntents are ac	ceptable to me and that I will act in accordance with its terms
Conor W	alsh, Ph.D.		

²⁰

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

EXHIBIT A

Research Plan

<u>Aim</u>: Harvard expects to perform research to support Company's development of both stroke and multiple sclerosis ("MS") specific Soft Exosuits as follows:

[**]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit B

Form of Harvard Invoice

Attached.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.



Email Address

Office for Sponsored Programs

SPONSOR NAME SPONSOR ADDRESS City, State Zip Line 3 Line 4 Line 5 HU PI:
Prime PI:
Award #:
Grant #:
Obligated \$:
Period:
PO #
Extra Header Field1

Extra Header Field2

PI Name
Prime PI Name
Award_No.
Grant_No.
\$0.00
00/00/0000 - 00/00/0000
PO_No.
misc header field1
misc header field2

Project Title: PROJECT TITLE

Invoice NumberInvoice DateTerms123456-01TBDNET 45

Amount Due Per Terms of Agreement: \$0.00

Space is provided here for a couple of lines of custom text, if required

Approved by: OSP Approver Name

Remittance stub if paying by check

Please indicate the HU Acct# 000-00000-000000 as reference with your payment.

Invoice NumberInvoice DateTerms123456-01TDBNET 45

PLEASE SEND REMITTANCE TO: President & Fellows of Harvard College P.O. Box 415649 Boston, MA 02241-5649 or

By WireTransfer/ACH Transfer to: Bank of America Harvard University Account #898-41099 (WIRE) ABA 011000138

(ACH) ABA 011000138 Swift Code: BOFAUS3N

For questions, please contact: OSP-FRAP@Harvard.edu

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit C

Harvard Background Patents

Case	Country	Туре	Appl. Title	Serial Number
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit D

Visitor Participation Agreement

Attached.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

HARVARD UNIVERSITY VISITOR PARTICIPATION AGREEMENT

for visitors from for-profit organizations (U.S. and foreign)

In consideration of my participation in research or other projects of President and Fellows of Harvard College ("Harvard" or the "University") and/or my use of funds, facilities or other resources provided by or through Harvard (collectively, "Harvard Activities"), I agree as follows:

- 1) With respect to my Harvard Activities, I acknowledge that I am bound by and will comply with Harvard's research policies (among them, the "Statement of Policy in Regard to Intellectual Property," or "IP Policy"), as they may be amended from time to time and made available by Harvard at http://vpr.harvard.edu/pages/research-policies-guidance or otherwise.
- 2) If any agreement between Harvard and a third party (*e.g.*, with respect to research funding or collaboration, or the use of materials, data, rights, equipment or the like) pertains to my Harvard Activities, I will abide that agreement, to the extent its terms apply and are made known to me.
- 3) I will report promptly to Harvard's Office of Technology Development ("OTD") any Invention, Sponsored Computer Software and/or Unpatented Material, as those capitalized terms are defined in the IP Policy, here referred to together as "Developments", that the IP Policy requires me to disclose to Harvard. I hereby assign and agree to assign to Harvard all my right, title and interest in all Developments that Harvard is entitled to own under the IP Policy, and all patent rights, copyrights and other intellectual property rights in those Developments throughout the world. I hereby grant to Harvard any rights required to be granted to Harvard under the IP Policy with respect to Incidental Inventions (as defined in the IP Policy) or other Developments I am entitled to own.
- 4) If any agreement between Harvard and a third party requires that my copyrights in works of authorship (for example, research reports or articles) be assigned or licensed to Harvard or a third party, I hereby assign or license to Harvard the copyrights that are needed in order to fulfill the agreement.
- 5) I will cooperate with Harvard in such reasonable steps as may be needed to carry out its research policies and to confirm, establish or protect the rights of Harvard or its designees. This may include, for example, executing additional documents, assisting in the filing, prosecution or defense of patent applications, and letting Harvard see my research data or materials as needed to respond to inquiries or conduct internal and external oversight activities.
- 6) I have not made with any third party, except the organization by which I am primarily employed and whose duly-authorized representative has signed this Agreement in the space provided below, any agreement that conflicts or reasonably could be construed as conflicting with the terms of this Agreement, and will not in future make any such agreement.
- 7) The obligations of this Agreement relating to my Harvard Activities will continue after the end of those Activities. This Agreement is binding on me, my estate, heirs and assigns.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

provided t	below, I hereby accept and agree to the terms and conditions of this Agreement:	
_		
_		
_		
rights in i	nventions, discoveries, copyrightable materials or other developments, which obligation is or reasonably could	d be construed
_		
_		
_		
on:		
_		
t	l below actrights in ite terms of tion.	below acknowledges and agrees that the individual whose signature appears above is released from any obligation to it rights in inventions, discoveries, copyrightable materials or other developments, which obligation is or reasonably could be terms of this Agreement. The person signing below represents and warrants that he/she is duly authorized to sign this Action.

<sup>27
*</sup> Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit E

Acknowledgment of Risk and Release

Attached.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

ACKNOWLEDGEMENT OF RISK AND RELEASE

for

Non-Harvard Personnel Using Harvard Research and Instructional Laboratory Facilities

THIS IS A RELEASE OF LEGAL RIGHTS - PLEASE READ AND UNDERSTAND BEFORE SIGNING

I, the undersigned, accept and agree to the following terms and conditions in consideration for my use of Harvard's research and instructional laboratory facilities.

- 1. Access to Facilities. The facilities are being made available to me as an educational or research opportunity. I am not a student, employee or affiliate of Harvard.
- 2. <u>Health and Safety Risks</u>. I understand that Harvard laboratories may contain hazardous substances and equipment. I will take every precaution necessary to protect my health and safety, and the health and safety of others. I will acquaint myself with and conduct my activities in accordance with all safety rules and safe operational procedures. If I am not familiar with or I do not know how to handle safely a substance or piece of equipment, I will seek assistance from qualified Harvard personnel. I recognize that I may be subjected to potential risks, illnesses and injuries. I have made my own investigation of these risks, understand these risks and assume them knowingly and willingly.
- 3. <u>No Medical Coverage</u>. I understand that if I am injured as a result of my activities at Harvard, I am not covered by Harvard insurance of any kind. It will be my responsibility to pay for emergency room care, doctors' services, hospitalization, and any other related costs, medical or non-medical. I will not be eligible to participate in Harvard's health, disability or life insurance program.
- 4. <u>Appropriate Conduct</u>. I agree to observe all applicable governmental, University and departmental policies, rules and regulations that pertain to my conduct on campus and in the facilities. I agree that Harvard officials may require me to leave the facilities if they believe that I have violated a policy, rule or regulation or if they believe that my conduct is inappropriate.
- 5. <u>Confidentiality.</u> I agree not to disclose or to use, directly or indirectly, any proprietary or confidential research, data, trade secrets, personal data, or other similar information of which I may become aware as a result of my activities in Harvard's facilities.
- 6. <u>ASSUMPTION OF RISK AND RELEASE OF CLAIMS</u>. Knowing the risks described above, I agree, on behalf of my family, heirs and personal representative(s), to assume all the risks and responsibilities surrounding my use of and access to Harvard's laboratories and equipment. To the maximum extent permitted by law, I release Harvard, its current and former members of its governing boards, officers, faculty, staff, representatives, volunteers, employees, students, other trainees and agents, and their respective heirs and assigns, from any and all claims, losses, expenses, damages, or liabilities which I may incur or suffer, arising out of or related to my use of- or access to the laboratories or equipment and resulting from any cause, including but not limited to negligence by Harvard, its current or former members of governing boards, officers, faculty, staff, representatives, volunteers, employees, students, other trainees or agents.
- 7. <u>INDEMNIFICATION</u>. I agree to indemnify Harvard, its current and former members of its governing boards officers, faculty, staff, representatives, volunteers, employees, students, other trainees, and agents, and their respective heirs and assigns, against any and all claims, actions, suits, procedures, costs, expenses, damages, and liabilities (including reasonable attorneys' fees) (collectively "Claims"), arising out of or related to my use of or access to Harvard's laboratories or equipment, but only in proportion to and to the extent that such Claims result from or are caused by my own negligent or intentional acts or omissions.

20

* Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Harvard University Visitor Participation Agreement.			
Signed:	Date:		
Name (print)			
Date of Arrival:	Anticipated Date of Departure:		
Version: 5/2015			

I have carefully read this Acknowledgement of Risk and Release before signing it. This agreement shall be governed by the laws of the Commonwealth of Massachusetts (excluding its conflict of laws principles), which shall be the forum for any lawsuits filed under or incident to this document or the attached

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit F

Press Release

Attached.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.



FOR IMMEDIATE RELEASE May 17, 2016

ReWalk Announces Collaboration with Harvard University's Wyss Institute to Develop Lightweight and Soft Exoskeleton Systems for the Treatment of Stroke, Multiple Sclerosis and Limited Mobility Patients

Five-Year Agreement Establishes Unique Collaboration and Extensive IP Licensing for "Soft Suit" Exoskeleton Systems

YOKNEAM ILIT, Israel and MARLBOROUGH, MA — ReWalk Robotics Ltd. (Nasdaq: RWLK) ("ReWalk" or "Company"), the leading global exoskeleton developer and manufacturer today announced a collaboration with Harvard University's Wyss Institute for Biologically Inspired Engineering for the licensing of Intellectual Property (IP), and development of concepts and designs of lightweight exoskeleton system technologies for lower limb disabilities. This exclusive licensing and collaboration agreement will focus on the development of "soft suit" systems for the treatment of stroke, Multiple Sclerosis (MS), mobility limitations for the elderly and other medical applications.

"There is a great need in the health care system for lightweight, lower-cost wearable exoskeleton designs to support stroke patients, individuals diagnosed with Multiple Sclerosis and senior citizens who require mechanical mobility assistance. This collaboration will help create the next generation of exoskeleton systems, making life-changing technology available to millions of consumers across a host of patient populations," said Larry Jasinski, CEO of ReWalk.

"This is a very exciting day for the soft suit technology," said Conor Walsh, who is a Core Faculty Member at the Wyss Institute, the John L. Loeb Associate Professor of Engineering and Applied Sciences at Harvard's John A. Paulson School of Engineering and Applied Sciences and Founder of the Harvard Biodesign Lab. "ReWalk brings commercialization expertise and experience in the area of wearable robotics and complements our translation-focused research. Ultimately this deal paves the way for this technology to make its way to patients."

The majority of Stroke and MS patients as well as the elderly do not require the structural support of current market rigid exoskeleton systems for individuals with spinal cord injury. The soft suit prototypes from the Wyss Institute transmit power to key joints of the legs with cable technologies powered with software and mechanics that are similar to the technologies used in the ReWalk system. The cables are connected to fabric-based designs that attach to the legs and foot, thus lending the name "soft suit."

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

"Our collaboration with ReWalk is a wonderful example of the Wyss Institute model in action," said Wyss Institute Founding Director Don Ingber, M.D., Ph.D. "We work with industry to help de-risk the technologies we develop, both technically and commercially, and thereby expedite their translation into real world applications."

Initial pilot studies with stroke patients run at Harvard's Wyss Institute in collaboration with faculty and researchers from Boston University have demonstrated the function of the soft suit exoskeleton technology. ReWalk will work in concert with the Wyss Institute on the continued development of lightweight designs to complete clinical studies, pursue regulatory approvals and commercialize the systems on a global scale. The first commercial application is expected to be stroke, followed by MS and then additional applications. There are an estimated 3 million stroke patients with lower limb disability in the U.S., and approximately 400,000 individuals with Multiple Sclerosis.

"Harvard and its Wyss Institute are pioneers in the development of technology in this space. The licensed Harvard patent portfolio currently includes 19 patent applications, which includes applications in at least six countries. The applications cover the soft suit, control systems and methods of treating patients. Harvard and the Wyss Institute have built comprehensive research expertise in addition to the worldwide patent portfolio. There is no better partner than these renowned institutions with which to pursue the mission of bringing cutting-edge technology to disabled individuals around the world," Jasinski added.

Coordinated by Harvard's Office of Technology Development, this collaboration includes funding for continued research and technology development at the Wyss Institute and transfer of knowledge and research results to ReWalk.

The agreement is effective May 16, 2016, with anticipated commercialization before 2019.

For more about Harvard's Wyss Institute, please visit: http://wyss.harvard.edu/

For more about Harvard's Office of Technology Development, please visit: http://otd.harvard.edu/

For more about ReWalk, please visit: www.rewalk.com

About ReWalk Personal 6.0

ReWalk Personal 6.0 is a wearable robotic exoskeleton that provides powered hip and knee motion to enable individuals with spinal cord injury to stand upright and walk. The system provides user-initiated mobility through the integration of a wearable brace support, a computer-based control system and motion sensors. The system allows independent, controlled walking similar to a natural gait pattern of the legs. The ReWalk device is the most studied exoskeleton in the industry

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

About ReWalk Robotics Ltd.

ReWalk Robotics Ltd. develops, manufactures and markets wearable robotic exoskeletons for individuals with spinal cord injury. Our mission is to fundamentally change the quality of life for individuals with lower limb disability through the creation and development of market leading robotic technologies. Founded in 2001, ReWalk has headquarters in the U.S., Israel and Germany. For more information on the ReWalk systems, please visit http://www.rewalk.com.

ReWalk® is a registered trademark of ReWalk Robotics Ltd. in Israel.

Forward Looking Statements

In addition to historical information, this press release contains forward-looking statements within the meaning of the U.S. Private Securities Litigation Reform Act of 1995, Section 27A of the U.S. Securities Act of 1933, and Section 21E of the U.S. Securities Exchange Act of 1934. Such forward-looking statements may include projections regarding ReWalk's future performance and, in some cases, may be identified by words like "anticipate," "assume," "believe," "continue," "could," "estimate," "expect," "intend," "may," "plan," "potential," "predict," "project," "future," "will," "seek" and similar terms or phrases. The forward-looking statements contained in this press release are based on management's current expectations, which are subject to uncertainty, risks and changes in circumstances that are difficult to predict and many of which are outside of ReWalk's control. Important factors that could cause ReWalk's actual results to differ materially from those indicated in the forward-looking statements include, among others: ReWalk's expectations regarding future growth, including its ability to increase sales in its existing geographic markets and to expand to new markets; ReWalk's ability to maintain and grow its reputation and to achieve and maintain market acceptance of its products; ReWalk's ability to achieve reimbursement from third-party payors for its products; ReWalk's expectations as to its clinical research program and clinical results; ReWalk's ability to improve its products, develop new products; ReWalk's ability to maintain adequate protection of its intellectual property and to avoid violation of the intellectual property rights of others; ReWalk's ability to repay its secured indebtedness; ReWalk's ability to gain and maintain regulatory approvals; ReWalk's ability to maintain relationships with existing customers and develop relationships with new customers; and other factors discussed under the heading "Risk Factors" in ReWalk's Annual Report on Form 10-K for the year ended December 31, 2015 filed with the U.S. Securities and Exchange Commission on February 29, 2016 and other documents subsequently filed with or furnished to the U.S. Securities and Exchange Commission. Any forward-looking statement made in this press release speaks only as of the date hereof. Factors or events that could cause ReWalk's actual results to differ from the statements contained herein may emerge from time to time, and it is not possible for ReWalk to predict all of them. Except as required by law, ReWalk undertakes no obligation to publicly update any forward-looking statements, whether as a result of new information, future developments or otherwise.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Media Contact:

ReWalk Robotics Jennifer Wlach 202-261-4000 jwlach@mercuryllc.com

Wyss Institute at Harvard University Mary Tolikas 617-432-7733 mary.tolikas@wyss.harvard.edu

Investor Relations Contact:

Lisa M. Wilson In-Site Communications, Inc. 917-543-9932 <a href="mailto:list.org/li

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

EXECUTION VERSION

LICENSE AGREEMENT

This License Agreement (together with all Exhibits hereto, this "Agreement") is entered into as of this 16th day of May, 2016 (the "Effective Date"), by and between President and Fellows of Harvard College, a charitable corporation of Massachusetts having an office at Richard A. and Susan F. Smith Campus Center, Suite 727, 1350 Massachusetts Avenue, Cambridge, Massachusetts 02138 ("Harvard"), and ReWalk Robotics, Ltd. a company existing under the laws of the State of Israel, having a place of business at 200 Donald Lynch Blvd., Marlborough, MA 01752 ("Licensee"). Harvard and Licensee each shall be referred to herein as a "Party" and together as the "Parties."

WHEREAS, certain technology claimed in the Licensed Patent Rights (as defined below) and included in the Licensed Know-How was developed in research conducted by Harvard researcher Dr. Conor Walsh, together with others at the Wyss Institute for Biologically Inspired Engineering at Harvard University; and

WHEREAS, Licensee wishes to obtain a license under the Licensed Patent Rights and Licensed Know-How; and

WHEREAS, Harvard desires to have products based on the inventions described in the Licensed Patent Rights developed and commercialized to benefit the public; and

WHEREAS, Licensee has represented to Harvard, in order to induce Harvard to enter into this Agreement, that Licensee shall commit itself to commercially reasonable efforts to develop, obtain regulatory approval for and commercialize such products in accordance with the terms and conditions set forth herein.

NOW, THEREFORE, the Parties hereto, intending to be legally bound, hereby agree as follows:

1. Definitions.

Whenever used in this Agreement with an initial capital letter, the terms defined in this Article 1, whether used in the singular or the plural, will have the meanings specified below. Terms with an initial capital letter used in this Agreement that are not defined below will have the meaning set forth in the Collaboration Agreement.

- **1.1** "Active" means, with respect to a Soft Exosuit, technology that commands motors to apply torques at joint(s) beyond adjusting pretension levels.
- **1.2** "Affiliate" means, with respect to a person, organization or entity, any person, organization or entity controlling, controlled by or under common control with, such person, organization or entity. For purposes of this definition only, "control" of another person, organization or entity will mean the possession, directly or indirectly, of the power to direct or cause the direction of the activities, management or policies of such person, organization or entity, whether through the ownership of voting securities, by contract or otherwise. Without limiting the foregoing, control will be presumed to exist when a person, organization or entity (a) owns or directly controls fifty percent (50%) or more of the outstanding voting stock or other ownership interest of the other organization or entity or (b) possesses, directly or indirectly, the power to elect or appoint fifty percent (50%) or more of the members of the governing body of the other organization or entity. The Parties acknowledge that in the case of certain entities organized under the laws of certain countries outside of the United States, the maximum percentage ownership permitted by law for a foreign investor may be less than fifty percent (50%), and that in such cases such lower percentage will be substituted in the preceding sentence.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.3** "Calendar Quarter" means each of the periods of three (3) consecutive calendar months ending on March 31, June 30, September 30 and December 31 during the Term.
- **1.4** "Clinical Data" means all records, reports, data and information and other Know-How concerning the Licensed Products that is or was created as a result of or in connection with the conduct of a clinical trial.
- **1.5** "Collaboration Agreement" means the research collaboration agreement entered into by the Parties as of the Effective Date under which the Parties will perform a Research Plan as defined therein.
- **1.6** "Consulting Invention" means any patentable invention (other than a Joint Consulting Invention) conceived and/or reduced to practice solely by Dr. Walsh within three (3) years after the Effective Date (so long as he is employed by Harvard during such three (3) year period) in his performance of consulting or other services for Licensee which is dominated by a Valid Claim within the Harvard Background Patents.
- **1.7** "Consulting Patent Rights" means any Patent Rights for any patent application or patent that claim any Consulting Invention, in each case solely to the extent the claims are directed to the subject matter of such Consulting Invention.
- **1.8** "Control" means, with respect to any Know-How, Patent Rights, other intellectual property rights or trade secrets, the legal authority or right (whether by license or otherwise) of a Party to grant a license or a sublicense of or under such Know-How, Patent Rights, or other intellectual property rights to another Person, or to otherwise disclose such trade secrets to another Person, without breaching the terms of any agreement with a third party, or misappropriating the trade secrets of a third party.
- **1.9** "**Design Freeze**" means the date that the design specifications of a Licensed Product have been achieved in accordance with the Licensee's quality system, which date shall be indicated in a written notice provided by Licensee to Harvard.
 - **1.10** "Developing Country" means any low-income or lower-middle-income country, as defined by the World Bank, except for India.
 - **1.11** "Development Milestones" means the development and commercialization milestones set forth in Exhibit 1.11 hereto.
- **1.12** "**Development Plan**" means the plan for the development and commercialization of Licensed Products attached hereto as <u>Exhibit 1.12</u>, as such plan may be adjusted from time to time pursuant to Section 3.2.
- 1.13 "Distributor" means a third party whom Licensee, its Affiliate or a Sublicensee engages to offer for sale, sell and/or import Licensed Products purchased from Licensee, such Affiliate or such Sublicensee, as applicable, for resale by such third party under the label of Licensee, such Affiliate or such Sublicensee, as applicable; provided that the term "Distributor" shall not include any person or entity who pays Licensee, its Affiliate or a Sublicensee any consideration (in any form) with respect to such engagement other than the consideration paid for the purchase of such Licensed Products.
- **1.14 "European Economic Area"** or **"EEA"** means the twenty-eight members of the European Union, Iceland, Liechtenstein, Norway, Switzerland and Turkey.
 - **1.15** "**Field**" means Active Medical Lower Limb Soft Exosuits.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.16** "First Commercial Sale" means the date of the first sale or lease by Licensee, its Affiliate or a Sublicensee of a Licensed Product to a third party for end use (or use by a lease) or consumption of such Licensed Product following receipt of any required Marketing Clearance in the country in which such Licensed Product is sold or leased, excluding, however, any sale or other distribution (including leases) for use in a clinical study.
- **1.17** "Governmental Authority" means any transnational, domestic or foreign federal, state or local governmental, regulatory or administrative authority, department, court, agency or official, including any political subdivision thereof.
- **1.18** "Harvard Background Patents" means, in each case to the extent owned or Controlled by Harvard as of the Effective Date, all of the Patent Rights for the patents and patent applications identified as "Harvard Background Patents" in <u>Exhibit 1.25</u>.
- **1.19** "Harvard Type A Invention" means any patentable invention that either (a) is conceived and reduced to practice (i) (x) in the performance of the Research Plan during the Research Term or (y) conceived in the performance of the Research Plan during the Term and reduced to practice within one (1) year after termination or expiration of the Research Term, (ii) for which at least one inventor is a member of the Harvard Team (but no inventor is a member of the Company Team) and (iii) which is dominated by a Valid Claim within the Harvard Background Patents (regardless of whether or not such Valid Claim ultimately issues); or (b) is deemed a Harvard Type A Invention pursuant to Section 5.2.4 of the Collaboration Agreement.
- **1.20** "Harvard Type A Patent Rights" means all of the Patent Rights for any patents or patent applications that claim any Harvard Type A Invention, in each case solely to the extent the claims are directed to the subject matter of such Harvard Type A Invention.
- **1.21** "Joint Consulting Invention" means any patentable invention conceived and/or reduced to practice jointly by (a) one or more employees or consultants of Licensee and (b) Dr. Walsh within [**] after the Effective Date (for so long as Dr. Walsh is employed by Harvard during such [**] period) in the performance of consulting or other services for Licensee which is dominated by a Valid Claim within the Harvard Background Patents.
- **1.22** "Joint Consulting Patent Rights" means any Patent Rights for any patent or patent application that claim any Joint Consulting Invention, in each case solely to the extent the claims are directed to the subject matter of such Joint Consulting Invention.
- 1.23 "Know-How" means all data, materials, compositions, methods, processes, analyses, formulae, know-how, trade secrets, unpatented inventions (whether patentable or unpatentable and whether or not reduced to practice), technology, and information generated in the performance of the Research, but excluding Inventions.
- **1.24** "**Licensed Know-How**" means all Know-How that is owned or Controlled by Harvard as of the Effective Date that (a) is set forth on Exhibit 1.24 hereto; or (b) the Parties both (i) mutually agree was known by a member of the Harvard Team or Dr. Walsh as of the Effective Date and is necessary for an individual who is reasonably skilled in the relevant art to enable him/her to practice the Harvard Background Patents; and (ii) mutually agree to add to Exhibit 1.24.
- **1.25** "**Licensed Patent Rights**" means (a) the Harvard Background Patents and (b) any Licensed Type A Patents, Consulting Patent Rights and Joint Consulting Patent Rights subsequently added to Exhibit 1.25 pursuant to Sections 2.5 or 2.6, as applicable.

* Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.26** "Licensed Product" means on a country-by-country basis, any product, the making, using, selling, offering for sale, importing or exporting in the country in question would (without the license granted hereunder) infringe directly, indirectly by inducement of infringement, or indirectly by contributory infringement, at least one pending Valid Claim or issued Valid Claim in that country.
- **1.27** "Marketing Clearance" means all approvals from the relevant Regulatory Authority of a country necessary to market and sell a Licensed Product in such country.
- **1.28** "Medical Lower Limb" means actively affecting leg movement in patients with gait impairments resulting from lack of proper muscle function due to neurological impairment, muscular degeneration or advanced aging.
- 1.29 "Net Sales" means the gross amount billed or invoiced by or on behalf of Licensee, its Affiliates or Sublicensees (in each case, the "Invoicing Entity") on sales, leases or other transfers of Licensed Products, less the following to the extent applicable with respect to such sales, leases or other transfers and not previously deducted from the gross invoice price: (a) customary trade, quantity or cash discounts to the extent actually allowed and taken; (b) amounts actually repaid or credited by reason of rejection or return of any previously sold, leased or otherwise transferred Licensed Products; (c) customer freight charges that are paid by or on behalf of the Invoicing Entity; and (d) to the extent separately stated on purchase orders, invoices or other documents of sale, any sales, value added or similar taxes, custom duties or other similar governmental charges levied directly on the production, sale, transportation, delivery or use of a Licensed Product that are paid by or on behalf of the Invoicing Entity, but not including any tax levied with respect to income; provided that:
- **1.29.1** in any transfers of Licensed Products between an Invoicing Entity and an Affiliate or Sublicensee of such Invoicing Entity not for the purpose of resale by such Affiliate or Sublicensee, Net Sales will be equal to the fair market value of the Licensed Products so transferred, assuming an arm's length transaction made in the ordinary course of business, and
- **1.29.2** in the event that an Invoicing Entity receives non-cash consideration for any Licensed Products or in the case of transactions not at arm's length with a non-Affiliate of an Invoicing Entity, Net Sales will be calculated based on the fair market value of such consideration or transaction, assuming an arm's length transaction made in the ordinary course of business.

Sales of Licensed Products by an Invoicing Entity to its Affiliate or a Sublicensee for resale by such Affiliate or Sublicensee will not be deemed Net Sales. Instead, Net Sales will be determined based on the gross amount billed or invoiced by such Affiliate or Sublicensee upon resale of such Licensed Products to a third party purchaser.

- **1.30** "Non-Royalty Sublicense Income" means any payments or other consideration that Licensee or any of its Affiliates receives in connection with a Sublicense, other than royalties based on Net Sales. If Licensee or its Affiliate receives non-cash consideration in connection with a Sublicense or in the case of transactions not at arm's length, Non-Royalty Sublicense Income will be calculated based on the fair market value of such consideration or transaction, at the time of the transaction, assuming an arm's length transaction made in the ordinary course of business.
- 1.31 "Patent Rights" means, with respect to any patents or patent applications, all of the following: (a) such patents and patent applications (including the PCT and/or U.S. utility application claiming priority to such application(s) that are filed on or before the one year conversion date of such application(s)); (b) any patent or patent application that claims priority to and is a divisional, continuation, reissue, renewal, reexamination, substitution or extension of any patent application identified in (a); (c) any patents issuing on any patent application identified in (a) or (b), including any reissues, renewals, reexaminations, substitutions or extensions thereof; (d) any claim of a continuation-in-part application or patent (including any reissues, renewals, reexaminations, substitutions or extensions thereof) that is entitled to the priority date of, and is directed specifically to subject matter specifically described in, at least one of the patents or patent applications identified in (a), (b) or (c); (e) any foreign counterpart (including PCTs) of any patent or patent application identified in (a), (b) or (c) or of the claims identified in (d); and (f) any supplementary protection certificates, pediatric exclusivity periods, any other patent term extensions and exclusivity periods and the like of any patents and patent applications identified in (a) through (e).

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **1.32** "**Person**" means an individual, corporation, partnership, limited liability company, association, trust or other entity or organization, including a Governmental Authority.
- **1.33** "**Regulatory Authority**" means any applicable government regulatory authority involved in granting approvals for the manufacturing and marketing of a Licensed Product, including the Food and Drug Administration in the United States and any foreign equivalent.
 - **1.34 "Research Term"** means the term of the Collaboration Agreement.
- **1.35** "**Soft Exosuit**" means assistive technology comprised of (i) a motor/gear-driven actuation system, (ii) Bowden-cable-based transmission, (iii) a textile-based human interface, (iv) wearable sensors and (v) algorithms to control assistance.
- **1.36** "Sublicense" means: (a) any right granted, license given or agreement entered into by Licensee to or with any other person or entity, under or with respect to or permitting any use or exploitation of any of the Licensed Patent Rights or otherwise permitting the development, manufacture, marketing, distribution, use and/or sale of Licensed Products; (b) any option or other right granted by Licensee to any other person or entity to negotiate for or receive any of the rights described under clause (a); or (c) any standstill or similar obligation undertaken by Licensee toward any other person or entity not to grant any of the rights described in clause (a) or (b) to any third party; in each case regardless of whether such grant of rights, license given or agreement entered into is referred to or is described as a sublicense.
 - **1.37 "Sublicensee"** means any person or entity granted a Sublicense.
 - **1.38** "**Territory**" means worldwide.
- **1.39 "Third Party Proposed Product**" means an actual or potential Licensed Product that is for an application or market segment for which Harvard reasonably believes a Licensed Product is not being actively developed and commercialized by Licensee, its Affiliates or any Sublicensee hereunder.
- 1.40 "Valid Claim" means, on a country-by-country basis: (a) a claim of an issued and unexpired patent within the Licensed Patent Rights that has not been (i) held permanently revoked, unenforceable, unpatentable or invalid by a decision of a court or governmental body of competent jurisdiction, unappealable or unappealed within the time allowed for appeal, (ii) rendered unenforceable through disclaimer or otherwise, (iii) abandoned or (iv) permanently lost through an interference or opposition proceeding without any right of appeal or review; or (b) a pending claim of a pending patent application within the Licensed Patent Rights that (i) has been asserted and continues to be prosecuted in good faith and (ii) has not been abandoned or finally rejected without the possibility of appeal or refiling.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

1.41 Additional Defined Terms. The following capitalized terms shall have the meanings ascribed to them in the following sections of this Agreement:

Abandoned Patent Rights	Section 6.3
Agreement	Preamble
Bankruptcy Code	Section 11.18
Challenging Party	Section 4.6
Challenge Proceeding	Section 4.6
Claims	Section 9.1
Effective Date	Preamble
Ex-Field Licensee	Section 11.17
Explanation	Section 3.4
[**]	Exhibit 11
Harvard	Preamble
Harvard Names	Section 11.3
Indemnitees	Section 9.1
Infringement	Section 7.1
Initial Deadline	Section 3.4.2
Licensed Type A Patents	Section 2.5
Licensee	Preamble
OTD	Section 2.6.4
Party or Parties	Preamble
Plan	Section 3.4
Reimbursed Costs	Section 11.17
Second Deadline	Section 3.4.4
Suit	Section 11.6
Term	Section 10.1
Third Deadline	Section 3.4.5

2. License.

- **2.1 License Grants.** Subject to the terms and conditions set forth in this Agreement, Harvard hereby grants to Licensee (a) an exclusive, royalty-bearing license under the Licensed Patent Rights (and (x) if an option for an exclusive license thereunder has been exercised in accordance with Section 6.2.3 of the Collaboration Agreement, under Harvard's interest in and to the Joint Patents Rights; and (y) if an option for an exclusive license thereunder has been exercised in accordance with Section 2.6.6, under Harvard's interest in and to the Joint Consulting Patents Rights) solely to make, have made, offer for sale, sell, have sold, repair, service and import Licensed Products solely within the Field and in the Territory; and (b) a non-exclusive, royalty-free license to use any Licensed Know-How both (i) for internal research purposes and (ii) to make, have made, offer for sale, sell, have sold, repair, service, and import Licensed Products solely within the Field in the Territory; provided, however, that:
- **2.1.1** Harvard retains the right, for itself and for other not-for-profit research organizations to practice the Licensed Patent Rights and use the Licensed Know-How, in each case, within the scope of the licenses granted above, solely for research, educational and scholarly purposes; and
- **2.1.2** the United States federal government retains rights in the Licensed Patent Rights and Licensed Know-How pursuant to 35 U.S.C. §§ 200-212 and 37 C.F.R. § 401 et seq., and any right granted in this Agreement greater than that permitted under 35 U.S.C. §§ 200-212 or 37 C.F.R. § 401 et seq. will be subject to modification as may be required to conform to the provisions of those statutes and regulations.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **2.2 Affiliates; Sublicensees; Distributors.** The license granted to Licensee under <u>Section 2.1</u> includes the right to have some or all of Licensee's rights or obligations under this Agreement exercised or performed by one or more of Licensee's Affiliates; Sublicensees or Distributors on Licensee's behalf; provided, however, that:
- **2.2.1** no such Affiliate, Sublicensee or Distributor shall be entitled to grant, directly or indirectly, to any third party any right of whatever nature under, or with respect to, or permitting any use or exploitation of, any of the Licensed Patent Rights or Licensed Know-How, including any right to develop, manufacture, market or sell Licensed Products; and
- **2.2.2** any act or omission taken or made by an Affiliate, Sublicensee or Distributor of Licensee under this Agreement will be deemed an act or omission by Licensee.

2.3 Sublicenses.

- **2.3.1 Sublicense Grant**. Licensee will be entitled to grant Sublicenses to third parties under the license granted pursuant to Section 2.1 subject to the terms of this Section 2.3. Any such Sublicense shall be on terms and conditions in compliance with and not inconsistent with the terms of this Agreement.
- **2.3.2 Sublicense Agreements.** Licensee shall grant Sublicenses pursuant to written agreements, which will be subject and subordinate to the terms and conditions of this Agreement. Such Sublicense agreements will contain, among other things, the following:
 - **2.3.2.1** all provisions necessary to ensure Licensee's ability to perform its obligations under this Agreement;
- **2.3.2.2** a section substantially the same as <u>Article 9</u> of this Agreement, which also will state that the Indemnitees (as defined in <u>Section 9.1</u>) are intended third party beneficiaries of such Sublicense agreement for the purpose of enforcing such indemnification;
- **2.3.2.3** a provision clarifying that, in the event of termination of the license set forth in <u>Section 2.1</u> (in whole or in part (e.g., termination in a particular country)), any existing Sublicense agreement shall terminate to the extent of such terminated license;
 - **2.3.2.4** a provision prohibiting the Sublicensee from sublicensing its rights under such Sublicense agreement; and
- **2.3.2.5** a provision prohibiting the Sublicensee from assigning the Sublicense agreement without the prior written consent of Harvard, except that Sublicensee may assign the Sublicense agreement to an affiliate of Sublicensee or to a successor in connection with the merger, consolidation or sale of all or substantially all of its assets or that portion of its business to which the Sublicense agreement relates; provided, however, that any permitted assignee agrees in writing to be bound by the terms of such Sublicense agreement.
- **2.3.3 Delivery of Sublicense Agreement.** Licensee shall furnish Harvard with a fully executed copy of any Sublicense agreement, promptly after its execution. Harvard shall keep all such copies in its confidential files and shall use them solely for the purpose of monitoring Licensee's and Sublicensees' compliance with their obligations hereunder and enforcing Harvard's rights under this Agreement.

7
* Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

2.3.4 Breach by Sublicensee. Licensee shall be responsible for any breach of a Sublicense agreement by any Sublicensee that results in a material breach of this Agreement. Licensee shall either (a) cure such breach in accordance with Section 10.2.2 of this Agreement or (b) enforce its rights by terminating such Sublicense agreement in accordance with the terms thereof.

2.4 Third Party Proposed Products.

- **2.4.1** If a Third Party makes a bona fide proposal to Harvard for developing a Third Party Proposed Product and Harvard is interested in having such Third Party Proposed Product developed and commercialized, Harvard may notify Licensee of the Third Party's proposal, and shall include in such notification non-confidential information regarding the Third Party proposal. Within [**] after the receipt of such notification from Harvard, Licensee shall notify Harvard whether it is interested in developing such Third Party Proposed Product. Licensee's decision shall be in its sole discretion.
- **2.4.2** If Licensee (in its sole discretion) notifies Harvard within such [**] period that it is interested in developing such Third Party Proposed Product, the Parties will negotiate in good faith and agree, during the [**] following such notification, upon a development plan with respect to such Third Party Proposed Product, which development plan will be similar to the Development Plan with respect to other Licensed Products developed by Licensee, subject to necessary adjustments, and will include reasonable milestones. If the Parties agree on such development plan and milestones within such [**] period, Licensee shall maintain its exclusive license(s) hereunder with respect to such Third Party Proposed Product, but shall be obligated (a) to use commercially reasonable efforts to develop and commercialize the Third Party Proposed Product in accordance with such new development plan and (b) to meet the milestones with respect to the Third Party Proposed Product.
- **2.4.3** If (a) License (in its sole discretion) notifies Harvard that it is not interested in developing such Third Party Proposed Product, or (b) the Parties do not agree on a development plan and milestones, then neither Harvard nor any other Person (other than Licensee) shall have the right to (nor license nor authorize any third Person to) make, have made, offer for sale, sell, have sold and/or import any such Third Party Proposed Product without the prior written consent of Licensee in its sole discretion.
- 2.5 Harvard Type A Patent Rights. Subject to the agreement of the inventors of the inventions claimed in the relevant Harvard Type A Patent Rights, upon receipt by Harvard of (a) a [**] dollar (\$[**]) license fee for worldwide rights under each family of Harvard Type A Patent Rights and (b) the written request of Licensee identifying the relevant family of Harvard Type A Patent Rights, the Parties shall immediately amend Exhibit 1.25 to add such Harvard Type A Patent Rights and upon such amendment, such Harvard Type A Patent Rights shall automatically (and without any additional action of the Parties) be deemed to be "Licensed Type A Patents" licensed pursuant to this Agreement by Harvard to Licensee on the same terms as those set forth herein, including, without limitation, subject to Section 11.17, reimbursement to Harvard for all documented, out-of-pocket expenses incurred by Harvard in connection with such Harvard Type A Patent Rights in accordance with Article 6.

2.6 Ownership of Consulting Inventions.

2.6.1 Dr. Walsh's consulting agreements for the performing of consulting or other services to Licensee shall comply with all applicable Harvard policies with respect thereto. Therefore the entire right, title and interest in and to each Consulting Invention, and all corresponding Consulting Patent Rights, will be owned solely by Harvard. The Parties and Dr. Walsh agree that Dr. Walsh shall assign his entire right, title and interest in any such Consulting Invention to Harvard.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **2.6.2** The entire right, title and interest in and to each Joint Consulting Invention, and all corresponding Joint Patent Rights, will be owned jointly by Licensee and Harvard. The parties and Dr. Walsh agree that Dr. Walsh shall assign his entire right, title and interest in any such Joint Consulting Invention to Harvard and Licensee jointly. Subject to the terms and conditions of this Agreement (including, without limitation, the exclusive license granted by Harvard to Licensee pursuant to Section 2.1 hereof), each Party shall be free to use and license its undivided share of any and all Joint Consulting Inventions or any and all Joint Patent Rights without having to (i) obtain the agreement or consent of the other Party, (ii) provide notice of such use or licensing to the other Party or (iii) make any accounting to the other Party for such use or licensing or any revenues or profits derived from such use or licensing.
- **2.6.3** All determinations of inventorship under this Agreement shall be made in accordance with United States patent law. In case of dispute over inventorship, a mutually acceptable outside patent counsel shall make the determination of the inventor(s) by applying the standards contained in United States patent law.
- **2.6.4** Dr. Walsh shall disclose to Licensee and Harvard's Office of Technology Development ("**OTD**") in a confidential writing the conception and/or reduction to practice of any Consulting Invention and Joint Consulting Invention promptly after he becomes aware thereof. Harvard shall disclose to Licensee in a confidential writing the conception and/or reduction to practice of any Consulting Invention and Joint Consulting Invention of which it becomes aware, promptly after OTD's receipt of an invention disclosure form from Dr. Walsh.
- **2.6.5** Any consulting or other agreement pursuant to which Dr. Walsh (for so long as he is an employee of Harvard) performs services for or on behalf of Licensee shall be consistent with and subordinate to the provisions of this <u>Section 2.6</u>. Any such consulting agreement shall require Dr. Walsh to assign his rights in Consulting Inventions and Joint Consulting Inventions, but not in other inventions, in a manner consistent with the provisions of this <u>Section 2.6</u>, and shall allow Dr. Walsh to make the disclosures contemplated by <u>Section 2.6.4</u>. In the case of any discrepancy between <u>Section 2.6</u> of this Agreement and any such consulting agreement, the terms of this Agreement shall prevail.
- **2.6.6** Solely at the option of Licensee and upon its written request (at Licensee's sole discretion), the Parties shall immediately amend Exhibit 1.25 to add each family of Consulting Patent Rights (or Harvard's interest in each family of Joint Consulting Patent Rights) that Licensee identifies and upon such amendment, such Consulting Patent Rights (or Harvard's interest in such Joint Consulting Patent Rights) shall automatically (and without any additional action of the Parties) be deemed to be "**Licensed Patents**" licensed pursuant to this Agreement by Harvard to Licensee on the same terms as set forth herein, including, without limitation, subject to Section 11.17, reimbursement to Harvard for all documented, out-of-pocket expenses incurred by Harvard in connection with such Consulting Patent Rights or Joint Consulting Patent Rights in accordance with Article 6.
- **2.7 No Other Grant of Rights**. Except as expressly provided herein, nothing in this Agreement will be construed to confer any ownership interest, license or other rights upon Licensee by implication, estoppel or otherwise as to any technology, intellectual property rights, products or biological materials of Harvard, or any other entity, regardless of whether such technology, intellectual property rights, products or biological materials are dominant, subordinate or otherwise related to any Licensed Patent Rights or Licensed Know-How.
- **2.8 Know-How Transfer.**Without limiting the Parties' obligations under the Collaboration Agreement, as soon as reasonably practicable after the Effective Date and in any event by no later than [**] after the Effective Date, Harvard will, at no additional cost or expense to Licensee, (a) transfer and deliver to Licensee tangible copies of all documents and materials contained within subpart (a) of the definition of Licensed Know-How and (b) upon reasonable request and during normal business hours, provide technical assistance and have regular knowledge transfer discussions reasonably sufficent to transfer the Licensed Know-How to Licensee time.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

3. Development and Commercialization.

- **3.1 Diligence.** Licensee shall use commercially reasonable efforts and shall cause its Sublicensees to use commercially reasonable efforts: (a) to develop Licensed Products in accordance with the Development Plan; (b) to introduce Licensed Products into the commercial market; and (c) to market Licensed Products following such introduction into the market. In addition, Licensee, by itself or through its Affiliates or Sublicensees, shall achieve each of the Development Milestones by the deadlines specified in Exhibit 1.11.
- **3.2 Adjustments of Development Plan**. Licensee will be entitled, from time to time, to make such adjustments to the then applicable Development Plan as Licensee believes, in its good faith judgment, are needed in order to improve Licensee's ability to meet the Development Milestones.
- **3.3 Reporting.** Within [**] after the end of each calendar year, Licensee shall furnish Harvard with a written report summarizing its, its Affiliates' and its Sublicensees' efforts during the prior year to develop and commercialize Licensed Products, including: (a) research and development activities; (b) commercialization efforts; and (c) marketing efforts. Each report must contain a sufficient level of detail for Harvard to assess whether Licensee is in compliance with its obligations under <u>Section 3.1</u> and a discussion of intended efforts for the then current year. Together with each report, Licensee shall provide Harvard with a copy of the then current Development Plan.

3.4 Failure to Meet Development Milestone; Opportunity to Cure.

- **3.4.1** If Licensee believes that it will not achieve a Development Milestone, it may notify Harvard in writing in advance of the relevant deadline. Licensee shall include with such notice (a) a reasonable explanation of the reasons for such failure ("**Explanation**") and (b) a reasonable, detailed, written plan for promptly achieving a reasonable extended deadline and/or amended milestone ("**Plan**").
- **3.4.2** If Licensee so notifies Harvard, but fails to provide Harvard with both an Explanation and Plan, then Licensee will have an additional [**] from such notice or until the original deadline of the relevant Development Milestone, whichever is later (the "**Initial Deadline**"), to meet such milestone. Licensee's failure to provide Harvard with both an Explanation and Plan by the Initial Deadline shall constitute a material breach of this Agreement and Harvard shall have the right to terminate this Agreement forthwith.
- **3.4.3** If Licensee so notifies Harvard and provides Harvard with an Explanation and Plan by the Initial Deadline, both of which are acceptable to Harvard in its reasonable discretion, then if Harvard agrees in writing or fails to provide written notice of its objection Licensee within [**] of the Initial Deadline, then Exhibit 1.11 will be amended automatically to incorporate the extended deadline and/or amended milestone set forth in the Plan.
- **3.4.4** If Licensee so notifies Harvard and provides Harvard with an Explanation and Plan, but the Explanation is not acceptable to Harvard in its reasonable discretion (e.g., Licensee asserts development preference for a non-Licensed Product) then Harvard will provide written notice of its objection to Licensee explaining why the Explanation is not acceptable within [**] of the Initial Deadline, and Licensee will have an additional [**] from receipt of Harvard's written notice or until the original deadline of the relevant Development Milestone, whichever is later (the "Second Deadline"), to meet such milestone or to persuade Harvard that the Explanation is reasonable. Licensee's failure to meet such milestone or to persuade Harvard that the Explanation is reasonable by the Second Deadline shall constitute a material breach of this Agreement and Harvard shall have the right to terminate this Agreement forthwith.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **3.4.5** If Licensee so notifies Harvard and provides Harvard with an Explanation and Plan, but the Plan is not acceptable to Harvard in its reasonable discretion, then Harvard will provide written notice of its objection to Licensee explaining why the Plan is not acceptable and providing Licensee with suggestions for an acceptable Plan within [**] of the Initial Deadline. Licensee will then have one opportunity to provide Harvard with a reasonably acceptable Plan (or to persuade Harvard that its Plan is reasonable) within [**] after receipt of Harvard's written notice and suggestions, during which time Harvard agrees to work in good faith with Licensee in its effort to develop a reasonably acceptable Plan. If, within such [**], Licensee provides Harvard with a reasonably acceptable Plan, then Exhibit 1.11 will be amended automatically to incorporate the extended deadline and/or amended milestone set forth in the Plan. If, within such [**], Licensee fails to provide a reasonably acceptable Plan, then Licensee will have an additional [**] or until the original deadline of the relevant Development Milestone, whichever is later (the "**Third Deadline**"), to meet such milestone. Licensee's failure to provide such acceptable Plan or meet such milestone by the Third Deadline shall constitute a material breach of this Agreement and Harvard shall have the right to terminate this Agreement forthwith.
- **3.4.6** For clarity, if Licensee fails to achieve a Development Milestone and did not avail itself of the procedure set forth in this Section 3.4, such failure shall constitute a material breach of this Agreement and Harvard shall have the right to terminate this Agreement forthwith.
- 3.5 Regulatory Filings. Licensee shall have the exclusive right to prepare and present all regulatory filings necessary or appropriate in any country and to obtain and maintain any regulatory approval (including Marketing Clearances) required to market Licensed Products in any such country. Licensee shall solely own all right, title and interest in and to all such regulatory approvals and filings. In the event of any undisputed termination of this Agreement (other than termination by Licensee pursuant to Section 10.2.2.1 for the uncured material breach of Harvard), upon receipt of a written request from Harvard, (a) Licensee shall promptly provide Harvard with the right to reference, cross-reference, review, have access to, incorporate and use all documents and other materials (other than Clinical Data) filed by or on behalf of Licensee and its Affiliates with any Regulatory Authority in furtherance of applications for regulatory approval in the relevant country with respect to Licensed Products ("Regulatory Documents"); (b) Harvard shall be entitled to freely use solely for research, educational and scholarly purposes any Regulatory Documents delivered by Licensee to Harvard pursuant to this Section 3.5; and (c) Harvard may grant third Persons the royalty-bearing right to use any Regulatory Documents (as well as, to the extent permitted by applicable law, any Clinical Data that has been depersonalized to the reasonable satisfaction of Licensee), provided that the Parties mutually agree on a reasonable allocation between Harvard and Licensee of all royalties paid by such third Persons as a result of the use of such Regulatory Documents.

4. Consideration for Grant of License.

4.1 License Fee. In partial consideration for the rights and licenses to the Licensed Patent Rights granted hereunder, within [**] after execution of the Agreement, Licensee shall pay Harvard a [**] U.S. Dollars (US \$[**]) license fee.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **4.2 Milestone Payments**. In partial consideration for the rights and licenses granted hereunder, Licensee shall pay Harvard the following milestone payments upon the first Licensed Product of Licensee to reach each of the following Development Milestones, regardless of whether such Development Milestone is achieved by Licensee, an Affiliate of Licensee or a Sublicensee:
 - **4.2.1** [**] U.S. Dollars (US \$[**]) upon [**];
 - **4.2.2** [**] U.S. Dollars (US \$[**]) upon [**];
 - **4.2.3** [**] U.S. Dollars (US \$[**]) upon [**].
- **4.2.4** Licensee shall notify Harvard in writing within [**] following the achievement of each milestone described above, and shall make the appropriate milestone payment within [**] after the achievement of such milestone.
- **4.3 Royalty on Net Sales.** In partial consideration for the rights and licenses granted hereunder, Licensee shall pay Harvard royalties on Net Sales as follows:
- **4.3.1** an amount equal to [**] percent ([**]%) of Net Sales until achieving cumulative aggregate Net Sales of [**] U.S. Dollars (US \$[**]);
- **4.3.2** an amount equal to [**] percent ([**]%) of Net Sales while the cumulative aggregate of Net Sales is greater than [**] U.S. Dollars (US \$[**]) but less than [**] U.S. Dollars (US \$[**]); and
- **4.3.3** an amount equal to [**] percent ([**]%) of Net Sales after achieving cumulative aggregate Net Sales of [**] U.S. Dollars (US \$[**]) until the expiration or termination of the Term.
- **4.4 Commercial Milestone**. In partial consideration for the rights and licenses granted hereunder, Licensee shall pay Harvard a commercial milestone payment in the amount of [**] U.S. Dollars (US \$[**]) when cumulative, gross, worldwide sales exceed [**] U.S. Dollars (US \$[**]).
- **4.5 Non-Royalty Sublicense Income.** In partial consideration for the rights and licenses granted hereunder, Licensee will pay Harvard an amount equal to [**] percent ([**]%) of all Non-Royalty Sublicense Income.
- **4.6 Patent Challenge.** If Licensee, its Affiliate or a Sublicensee ("Challenging Party") commences an action in which it challenges the validity, enforceability or scope of any of the Licensed Patent Rights (a "Challenge Proceeding"), the royalty rates specified in Section 4.3 will be doubled with respect to Net Sales of Licensed Products that are sold during the pendency of such Challenge Proceeding. If the outcome of such Challenge Proceeding is a determination against the Challenging Party, (a) the royalty rate specified in Section 4.3 with respect to Net Sales of Licensed Products that are covered by the Licensed Patent Rights that are the subject of such Challenge Proceeding shall remain at such doubled rate and (b) Licensee shall reimburse Harvard for all expenses incurred by Harvard (including reasonable attorneys' fees) in connection with such Challenge Proceeding. If the outcome of such Challenge Proceeding is a determination in favor of the Challenging Party, Harvard shall reimburse Licensee [**] of all royalties paid by Licensee to Harvard during the pendency of such Challenge Proceeding and the royalty rate specified in Section 4.3 with respect to Net Sales of Licensed Products that are covered by the Licensed Patent Rights that are the subject of such Challenge Proceeding shall revert to the original (i.e., not doubled) rate.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

5. Reports; Payments; Records.

5.1 Reports and Payments.

- **5.1.1 Reports.** Within [**] after the conclusion of each Calendar Quarter commencing with the first Calendar Quarter in which Net Sales are generated or Non-Royalty Sublicense Income is received, Licensee shall deliver to Harvard a report containing the following information (in each instance, with a Licensed Product-by-Licensed Product and country-by-country breakdown):
- **5.1.1.1** the number of units of Licensed Products sold, leased or otherwise transferred by Invoicing Entities for the applicable Calendar Quarter;
- **5.1.1.2** the gross amount billed or invoiced for Licensed Products sold, leased or otherwise transferred by Invoicing Entities during the applicable Calendar Quarter;
 - **5.1.1.3** a calculation of Net Sales for the applicable Calendar Quarter, including an itemized listing of allowable deductions;
 - **5.1.1.4** a detailed accounting of all Non-Royalty Sublicense Income received during the applicable Calendar Quarter;
- **5.1.1.5** the total amount payable to Harvard in U.S. Dollars on Net Sales and Non-Royalty Sublicense Income for the applicable Calendar Quarter, together with the exchange rates used for conversion;
- 5.1.1.6 a list of Harvard Case numbers for all Licensed Patent Rights that have Valid Claims covering the Licensed Products; and
 - **5.1.1.7** A list of Licensed Products which incorporate or comprise the Licensed Know-How.

Each such quarterly report shall be certified on behalf of Licensee by its chief financial officer as true, correct and complete in all material respects. If no amounts are due to Harvard for a particular Calendar Quarter, the report shall so state. In addition to the above-described quarterly reports, within sixty (60) days after the end of each calendar year, Licensee shall provide Harvard with an annual report containing the above-described information for such calendar year. Each such annual report shall be prepared on behalf of Licensee by an independent, certified public accountant and certified by such accountant as true, correct and complete in all material respects.

- **5.1.2 Payment**. Within [**] after the end of each Calendar Quarter, Licensee shall pay Harvard all amounts due with respect to Net Sales and Non-Royalty Sublicense Income for the applicable Calendar Quarter.
- **5.2 Payment Currency**. All payments due under this Agreement will be paid in U.S. Dollars. Conversion of foreign currency to U.S. Dollars will be made at the conversion rate existing in the United States (as reported in the *Wall Street Journal*) on the last working day of the applicable Calendar Quarter. Such payments due will be paid without deduction of exchange, collection or other charges.

5.3 Records.

5.3.1 Licensee Records. Licensee shall maintain, and shall cause its Affiliates and Sublicensees to maintain, complete and accurate records of Licensed Products that are made, used, sold, leased or transferred under this Agreement, any amounts payable to Harvard in relation to such Licensed Products, and all Non-Royalty Sublicense Income received by Licensee and its Affiliates, which records shall contain sufficient information to permit Harvard to confirm the accuracy of any reports or notifications delivered to Harvard under Section 5.1. Licensee, its Affiliates and/or its Sublicensees, as applicable, shall retain such records relating to a given Calendar Quarter for at least [**] after the conclusion of that Calendar Quarter, during which time Harvard will have the right, at its expense, to cause an independent, certified public accountant reasonably acceptable to Licensee to inspect such records during normal business hours for the purposes of verifying the accuracy of any reports and payments delivered under this Agreement and Licensee's compliance with the terms hereof. Such accountant shall not disclose to Harvard any information other than information necessary to confirm the accuracy of reports and payments delivered under this Agreement. The Parties shall reconcile any underpayment or overpayment within [**] after the accountant delivers the results of the audit. If any audit performed under this Section 5.3 reveals an underpayment in excess of [**] percent ([**]%) in any calendar year, Licensee shall reimburse Harvard for all amounts incurred in connection with such audit. Harvard may exercise its rights under this Section 5.3 [**] every year per audited entity and only with at least [**] prior written notice to the audited entity.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **5.3.2 Harvard Records**. Harvard shall maintain complete and accurate records of its costs and expenses incurred under this Agreement, which records shall contain sufficient information to permit Licensee to confirm the accuracy of any reimbursable expenses paid by Licensee hereunder. Harvard shall retain such records relating to a given Calendar Quarter for at least [**] after the conclusion of that Calendar Quarter, during which time Licensee will have the right, at its expense, to cause an independent, certified public accountant reasonably acceptable to Harvard to inspect such records during normal business hours for the purposes of verifying the accuracy of any invoices delivered under this Agreement and Harvard's compliance with the terms hereof. Such accountant shall not disclose to Licensee any information other than information necessary to confirm the accuracy of such invoices and the reimbursable costs and expenses claimed by Harvard under this Agreement. The Parties shall reconcile any underpayment or overpayment within [**] after the accountant delivers the results of the audit. If any audit performed under this Section 5.3.2 reveals an overpayment by Licensee in excess of [**] percent ([**]%) in any calendar year, Harvard shall reimburse Licensee for all amounts incurred in connection with such audit. Licensee may exercise its rights under this Section 5.3.2 [**] every year and only with at least [**] prior written notice to Harvard.
- **5.4 Late Payments.** Any payments by Licensee that are not paid on or before the date such payments are due under this Agreement will bear interest at the lower of (a) [**] percent ([**]%) per month and (b) the maximum rate allowed by law. Interest will accrue beginning on the first day following the due date for payment and will be compounded quarterly.
- **5.5 Payment Method**. Each payment due to Harvard under this Agreement shall be paid by check or wire transfer of funds to Harvard's account in accordance with written instructions provided by Harvard. If made by wire transfer, such payments shall be marked so as to refer to this Agreement.
- **5.6 Withholding and Similar Taxes**. Licensee shall be responsible for all non-U.S. taxes related to payments due to Harvard hereunder. All such payments due to Harvard will be paid without deduction of such taxes.

* Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

6. Patent Filing, Prosecution and Maintenance.

6.1 Control.

- **Control by Harvard**. Harvard will be responsible for the preparation, filing, prosecution, protection, defense and maintenance of all Licensed Patent Rights, using independent patent counsel reasonably acceptable to Licensee. Harvard will: (i) instruct such patent counsel to furnish the Licensee with copies of all correspondence relating to the Licensed Patent Rights from the United States Patent and Trademark Office (USPTO) and any other patent office, as well as copies of all proposed responses to such correspondence in time for Licensee to review and comment on such response; (ii) give Licensee an opportunity to review the text of each patent application before filing; (iii) consult with Licensee with respect thereto; (iv) supply Licensee with a copy of the application as filed, together with notice of its filing date and serial number; and (v) keep Licensee advised of the status of actual and prospective patent filings. Harvard shall give Licensee the opportunity to provide comments on and make requests of Harvard concerning the preparation, filing, prosecution, protection, defense and maintenance of the Licensed Patent Rights, and shall seriously consider such comments and requests; however, final decision-making authority shall vest in Harvard (except where Licensee is responsible pursuant to Section 6.1(b)). Notwithstanding the foregoing, if Licensee requests that Harvard prepare, file and/or prosecute (or continue to prepare, file and/or prosecute) an application for any Licensed Patent Rights, then Harvard shall prepare, file and prosecute such application (or continue to prepare, file and/or prosecute) such application subject to Company's reimbursement to Harvard for all documented, out-of-pocket expenses incurred by Harvard in connection therewith in accordance with Article 6 (but Section 11.17), provided, however, that Harvard expressly reserves the right to decline Licensee's request to file, prosecute, maintain or defend any of the Licensed Patent Rights in any Developing Country(ies) unless (A) Licensee demonstrates to Harvard's reasonable satisfaction that the filing, prosecution, maintenance or defense of such Licensed Patent Rights in such Developing Country(ies) would materially increase the locally-affordable availability of Licensed Products or equivalents thereof (e.g., generic products) in those and/or other Developing Country(ies) and (B) the provisions of Section 7 notwithstanding, Licensee agrees that Harvard shall hold final decision-making authority, on a case-by-case basis, as to whether Licensee will be permitted to enforce such Licensed Patent Rights in such Developing Country(ies).
- **(b)** So long as Licensee is performing its obligation to reimburse Harvard for all documented, out-of-pocket expenses incurred by Harvard in connection with such Licensed Patent Rights in accordance with Article 6 (but subject to Section 11.17), Harvard will not abandon, withdraw, surrender or otherwise seek to no longer be responsible for the preparation, filing, prosecution, protection, defense and maintenance of any Licensed Patent Rights.
- **6.2 Expenses.** Subject to Sections 6.3 and 11.20, below, Licensee shall reimburse Harvard for all documented, out-of-pocket expenses incurred by Harvard pursuant to this Article 6 within [**] after the date of each valid and accurate invoice from Harvard for such expenses. In addition, within [**] after the Effective Date and subject to Section 11.17, Licensee shall reimburse Harvard for all documented, out-of-pocket expenses incurred by Harvard prior to the Effective Date with respect to the preparation, filing, prosecution, protection, and maintenance of the Harvard Background Patents, the billed amount of which is, as of the Effective Date, equal to [**] United States Dollars (US \$[**]). License shall pay such amount as follows: (a) [**] United States Dollars (US \$[**]) to be paid within [**] of the Effective Date and (b) the remaining balance to be paid on or before [**].
- **6.3 Abandonment.** If Licensee decides that it does not wish to pay for the preparation, filing, prosecution, protection, maintenance or defense of any Licensed Patent Rights (other than Joint Patent Rights) in a particular country ("**Abandoned Patent Rights**"), Licensee shall provide Harvard with prompt written notice of such election. Upon receipt of such notice by Harvard, Licensee shall be released from its obligation to reimburse Harvard for the expenses incurred thereafter as to such Abandoned Patent Rights; provided, however, that expenses authorized prior to the receipt by Harvard of such notice shall be deemed incurred prior to the notice. In the event of Licensee's abandonment of any Licensed Patent Rights, any license granted by Harvard to Licensee hereunder with respect to such Abandoned Patent Rights will terminate, and Licensee will have no rights whatsoever to exploit such Abandoned Patent Rights. Harvard will then be free, without further notice or obligation to Licensee, to grant rights in and to such Abandoned Patent Rights to third parties.

¹⁵

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **6.4 Small Entity Designation**. If Licensee, its Affiliates, any Sublicensee and/or any holder of an option to obtain a Sublicense does not qualify, or at any point during the Term ceases to qualify, as an entity entitled to pay lesser fees as provided by the USPTO (i.e., a "small entity") or the patent office of any other country, Licensee shall so notify Harvard immediately, in order to enable Harvard to comply with regulations regarding payment of fees with respect to Licensed Patent Rights.
- **6.5 Marking.** Licensee shall, and shall cause its Affiliates and Sublicensees to, mark all Licensed Products sold or otherwise disposed of in such a manner as to conform with the patent laws and practice of the country to which such products are shipped or in which such products are sold for purposes of ensuring maximum enforceability of Licensed Patent Rights in such country.

7. Enforcement of Patent Rights.

- **7.1 Notice.** In the event either party becomes aware of any possible or actual infringement of any Licensed Patent Rights with respect to Licensed Products in the Field (an "**Infringement**"), that Party shall promptly notify the other Party and provide it with details regarding such Infringement.
- Suit by Licensee. Licensee shall have the first right, but not the obligation, to take action in the prosecution, prevention, or termination of any Infringement. Before Licensee commences an action with respect to any Infringement, Licensee shall consider in good faith the views of Harvard and potential effects on the public interest in making its decision whether to sue, especially with regard to the locally-affordable availability of Licensed Products or equivalents thereof, e.g., generic products, in Developing Countries, provided, however, that Licensee shall not be in breach of this Agreement if Licensee makes a commercially reasonable decision to take or not take action. Should Licensee elect to bring suit against an infringer, Licensee shall keep Harvard reasonably informed of the progress of the action and shall give Harvard a reasonable opportunity in advance to consult with Licensee and offer its views about major decisions affecting the litigation. Licensee shall give careful consideration to those views, but shall have the right to control the action; provided, however, that if Licensee fails to defend in good faith the validity and/or enforceability of the Licensed Patent Rights in the action or, or if Licensee's license to a Valid Claim in the suit terminates, Harvard may elect to take control of the action pursuant to Section 7.3. Any and all expenses, including reasonable attorneys' fees, incurred by Harvard with respect to the prosecution, adjudication and/or settlement of such suit, including any related appeals, shall be paid for entirely by Licensee and Licensee shall hold Harvard free, clear and harmless from and against any and all such expenses. The expenses of such suit or suits that Licensee elects to bring, including any expenses of Harvard incurred in conjunction with the prosecution of such suits or the settlement thereof, shall be paid for entirely by Licensee and Licensee shall hold Harvard free, clear and harmless from and against any and all costs of such litigation, including reasonable attorneys' fees. Licensee shall not compromise or settle such litigation without the prior written consent of Harvard, which consent shall not be unreasonably withheld or delayed. In the event Licensee exercises its right to sue pursuant to this Section 7.2, it shall first reimburse itself out of any sums recovered in such suit or in settlement thereof for all costs and expenses of every kind and character, including reasonable attorneys' fees, necessarily incurred in the prosecution of any such suit. If, after such reimbursement, any funds shall remain from said recovery, then Harvard shall receive an amount equal to [**] percent ([**]%) of such funds and the remaining [**] percent ([**]%) of such funds shall be retained by Licensee.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **Suit by Harvard.** If Licensee does not take action in the prosecution, prevention, or termination of any Infringement pursuant to Section 7.2 above, and has not commenced negotiations with the infringer for the discontinuance of said Infringement, within [**] after receipt of notice to Licensee by Harvard of the existence of an Infringement, Harvard may elect to do so. Should Harvard elect to bring suit against an infringer and Licensee is joined as party plaintiff in any such suit, Licensee shall have the right to approve the counsel selected by Harvard to represent Harvard and Licensee, such approval not to be unreasonably withheld. Any and all expenses, including reasonable attorneys' fees, incurred by Licensee with respect to the prosecution, adjudication and/or settlement of such suit, including any related appeals, shall be paid for entirely by Harvard and Harvard shall hold Licensee free, clear and harmless from and against any and all such expenses. Harvard shall not compromise or settle such litigation without the prior written consent of Licensee, which consent shall not be unreasonably withheld or delayed. In the event Harvard exercises its right to sue pursuant to this Section 7.3, it shall first reimburse itself out of any sums recovered in such suit or in settlement thereof for all costs and expenses of every kind and character, including reasonable attorneys' fees, necessarily incurred in the prosecution of any such suit. If, after such reimbursement, any funds shall remain from said recovery, then Licensee shall receive an amount equal to [**] percent ([**]) of such funds and the remaining [**] percent ([**]) of such funds shall be retained by Harvard.
- **7.4 Own Counsel**. Each Party shall always have the right to be represented by counsel of its own selection and at its own expense in any suit instituted under this Article 7 by the other Party for Infringement.
- **7.5 Cooperation.** Each Party agrees to cooperate fully in any action under this <u>Article 7</u> that is controlled by the other Party, including joining as a party to the action, provided that the controlling Party reimburses the cooperating Party promptly for any costs and expenses incurred by the cooperating Party in connection with providing such assistance.
- **7.6 Declaratory Judgment.** If a declaratory judgment action is brought naming Licensee and/or any of its Affiliates or Sublicensees as a defendant and alleging invalidity or unenforceability of any claims within the Licensed Patent Rights, Licensee shall promptly notify Harvard in writing and Harvard may elect, upon written notice to Licensee within [**] after Harvard receives notice of the commencement of such action, to take over the sole defense of the invalidity and/or unenforceability aspect of the action at its own expense.
- 8. Warranties; Limitation of Liability.
 - 8.1 Representations and Warranties.
 - **8.1.1 Reciprocal Representations and Warranties.** Each Party represents and warrants to the other Party as follows:
- **(a) Duly Organized.** Such Party is a corporation duly organized, validly existing and in good standing under the laws of the jurisdiction of its incorporation, is qualified to do business and is in good standing as a foreign corporation in each jurisdiction in which the conduct of its business or the ownership of its properties requires such qualification and failure to have such would prevent such Party from performing its obligations under this Agreement.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **(b) Due Authorization; Binding Agreement.** The execution, delivery and performance of this Agreement by such Party have been duly authorized by all necessary corporate action. This Agreement is a legal and valid obligation binding on such Party and enforceable in accordance with its terms and does not: (i) to such Party's knowledge and belief, violate any law, rule, regulation, order, writ, judgment, decree, determination or award of any court, governmental body or administrative or other agency having jurisdiction over such Party; nor (ii) conflict with, or constitute a default under, any agreement, instrument or understanding, oral or written, to which such Party is a party or by which it is bound.
- **8.1.2 Representations by Licensee.** Licensee represents and warrants that it will comply in all material respects, and will ensure that its Affiliates and Sublicensees comply in all material respects, with all local, state, federal and international laws and regulations relating to the development, manufacture, use, sale and importation of Licensed Products. Without limiting the foregoing, Licensee represents and warrants, on behalf of itself and its Affiliates and Sublicensees, that it shall comply with all United States laws and regulations controlling the export of certain commodities and technical data, including without limitation all Export Administration Regulations of the United States Department of Commerce. Among other things, these laws and regulations prohibit or require a license for the export of certain types of commodities and technical data to specified countries. Licensee hereby gives written assurance that it will comply with, and will cause its Affiliates to comply with (and will contractually obligate its Sublicensees to comply with), all United States export control laws and regulations, that it bears sole responsibility for any violation of such laws and regulations by itself or its Affiliates or Sublicensees, and that it will indemnify, defend, and hold Harvard harmless (in accordance with Section 9.1) for the consequences of any such violation.
 - **8.1.3 Representations by Harvard.** OTD, on behalf of Harvard, represents, warrants and covenants to Licensee as follows:
- (a) Third Party Rights. As of the Effective Date, to the knowledge of OTD, Harvard has not received any funding, participation or other rights from any third Person or other Governmental Authority that conflicts or is inconsistent with the rights, licenses and options granted to Licensee hereunder. As of the Effective Date, Harvard has the right to grant the licenses and rights as contemplated under this Agreement and has not granted any right to any third Person which would conflict or be inconsistent with the licenses and rights granted to Licensee hereunder. Harvard will not during the Term grant any right to any third Person which would conflict or be inconsistent with the licenses and rights granted to Licensee hereunder.
- **(b) Patents.** Harvard is the sole and exclusive owner of all right, title and interest in and to the Harvard Background Patents, including all of the patents and patent applications identified as "Harvard Background Patents" in Exhibit 1.25. All official fees, maintenance fees and annuities for the Licensed Patent Rights have been paid through the Effective Date.
- (c) Non-Infringement of Third Party Rights. OTD has not received any written notice from any Person of any actual or threatened claim that the use or practice of the Licensed Patent Rights or Licensed Know-How infringes or otherwise violates the intellectual property rights of a third Person.
- (d) Patent and Technology Status. As of the Effective Date, none of the Licensed Patent Rights is currently involved in any interference, reissue, reexamination, or opposition proceeding and OTD has not received any written notice from any Person of such actual or threatened proceeding.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

(e) Non-Infringement by Third Parties. As of the Effective Date, to the knowledge of OTD, no third Persons are infringing the Licensed Patent Rights or Licensed Know-How.

8.2 No Other Warranties.

- **8.2.1** NOTHING CONTAINED HEREIN SHALL BE DEEMED TO BE A WARRANTY BY HARVARD THAT PATENTS WILL ISSUE ON PATENT APPLICATIONS INCLUDED IN THE LICENSED PATENT RIGHTS, OR THAT ANY OF THE LICENSED PATENT RIGHTS WILL AFFORD ADEQUATE OR COMMERCIALLY WORTHWHILE PROTECTION.
- **8.2.2** HARVARD MAKES NO WARRANTIES WHATSOEVER AS TO THE COMMERCIAL OR SCIENTIFIC VALUE OF THE LICENSED PATENT RIGHTS OR LICENSED KNOW-HOW. HARVARD MAKES NO REPRESENTATION THAT THE PRACTICE OF THE LICENSED PATENT RIGHTS OR THE DEVELOPMENT, MANUFACTURE, USE, SALE OR IMPORTATION OF ANY LICENSED PRODUCT, OR ANY ELEMENT THEREOF, WILL NOT INFRINGE ANY PATENT OR PROPRIETARY RIGHTS.
- **8.2.3** EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, NEITHER PARTY MAKES ANY WARRANTY WITH RESPECT TO ANY TECHNOLOGY, PATENTS, GOODS, SERVICES, RIGHTS OR OTHER SUBJECT MATTER OF THIS AGREEMENT AND EACH PARTY HEREBY DISCLAIMS WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE AND NONINFRINGEMENT WITH RESPECT TO ANY AND ALL OF THE FOREGOING.

8.3 Limitation of Liability.

- **8.3.1** Except with respect to matters for which Licensee is obligated to indemnify Harvard under <u>Article 9</u>, neither Party will be liable to the other with respect to any subject matter of this Agreement under any contract, negligence, strict liability or other legal or equitable theory for (a) any indirect, incidental, consequential or punitive damages or lost profits or (b) cost of procurement of substitute goods, technology or services.
- **8.3.2** Except with respect to matters for which Licensee is obligated to indemnify Harvard under Article 9, each Party's aggregate liability for all damages of any kind arising out of or relating to this Agreement or its subject matter under any contract, negligence, strict liability or other legal or equitable theory shall not exceed the amounts paid by Licensee to Harvard under this Agreement.

9. Indemnification and Insurance.

9.1 Indemnity.

9.1.1 Indemnity. Licensee shall indemnify, defend and hold harmless Harvard and its current and former directors, governing board members, trustees, officers, faculty, medical and professional staff, employees, students, and agents and their respective successors, heirs and assigns (collectively, the "**Indemnitees**") from and against any third party claim, liability, cost, expense, damage, deficiency, loss or obligation of any kind or nature (including, without limitation, reasonable attorneys' fees and other costs and expenses of litigation) based upon, arising out of, or otherwise relating to the rights or licenses granted to Licensee under this Agreement or any Sublicense, including any cause of action relating to product liability concerning any product, process, or service made, used, sold or performed pursuant to any right or license granted under this Agreement (collectively, "**Claims**"). Neither Licensee nor Harvard shall settle any Claim without the prior written consent of the other, which consent shall not be unreasonably withheld.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

9.2 Insurance.

- **9.2.1** Beginning at the time any Licensed Product is being commercially distributed or sold (other than for the purpose of obtaining regulatory approvals) by Licensee, or by an Affiliate, Sublicensee or agent of Licensee, Licensee shall, at its sole cost and expense, procure and maintain commercial general liability insurance in amounts not less than [**] U.S. Dollars (US \$[**]) per incident and [**] U.S. Dollars (US \$[**]) annual aggregate and naming the Indemnitees as additional insureds. During clinical trials of any such Licensed Product, Licensee shall, at its sole cost and expense, procure and maintain commercial general liability insurance in such equal or lesser amount as Harvard shall require, naming the Indemnitees as additional insureds. Such commercial general liability insurance shall provide: (a) product liability coverage and (b) broad form contractual liability coverage for Licensee's indemnification obligations under this Agreement.
- **9.2.2** If Licensee elects to self-insure all or part of the limits described above in Section 9.2.1 (including deductibles or retentions that are in excess of [**] (US \$[**]) annual aggregate) such self-insurance program must be acceptable to Harvard and CRICO/RMF (Harvard's insurer) in their sole discretion. The minimum amounts of insurance coverage required shall not be construed to create a limit of Licensee's liability with respect to its indemnification obligations under this Agreement.
- **9.2.3** Licensee shall provide Harvard with written evidence of such insurance upon request of Harvard. Licensee shall provide Harvard with written notice at least fifteen (15) days prior to the cancellation, non-renewal or material change in such insurance. If Licensee does not obtain replacement insurance providing comparable coverage within such fifteen (15) day period, Harvard shall have the right to terminate this Agreement effective at the end of such fifteen (15) day period without notice or any additional waiting periods.
- **9.2.4** Licensee shall maintain such commercial general liability insurance beyond the expiration or termination of this Agreement during: (a) the period that any Licensee Product is being commercially distributed or sold by Licensee, or an Affiliate, Sublicensee or agent of Licensee; and (b) a reasonable period after the period referred to in (a) above which in no event shall be less than fifteen (15) years.

10. Term and Termination.

10.1 Term. The term of this Agreement shall commence on the Effective Date and, unless earlier terminated as provided in this <u>Article 10</u>, shall continue in full force and effect until the expiration of the last to expire Valid Claim (the "**Term**").

10.2 Termination.

10.2.1 Termination Without Cause. Licensee may terminate this Agreement for any or no reason upon sixty (60) days prior written notice to Harvard.

10.2.2 Termination for Default.

- **10.2.2.1** In the event that either Party commits a material breach of its obligations under this Agreement and fails to cure that breach within thirty (30) days after receiving written notice thereof, the other Party may terminate this Agreement immediately upon written notice to the Party in breach.
- **10.2.2.2** If Licensee defaults in its obligations under <u>Section 9.2</u> to procure and maintain insurance or, if Licensee has in any event failed to comply with the notice requirements contained therein, then Harvard may terminate this Agreement immediately without notice or additional waiting period.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

10.2.3 Bankruptcy. Harvard may terminate this Agreement upon notice to Licensee if Licensee becomes insolvent, is adjudged bankrupt, applies for judicial or extra-judicial settlement with its creditors, makes an assignment for the benefit of its creditors, voluntarily files for bankruptcy or has a receiver or trustee (or the like) in bankruptcy appointed by reason of its insolvency, or in the event an involuntary bankruptcy action is filed against Licensee and not dismissed within ninety (90) days, or if Licensee becomes the subject of liquidation or dissolution proceedings or otherwise discontinues business.

10.3 Effect of Termination.

- 10.3.1 Termination of Rights. Upon expiration or termination of this Agreement by either Party pursuant to any of the provisions of Section 10.2: (a) the rights and licenses granted to Licensee under Article 2 shall terminate, all rights in and to and under the Licensed Patent Rights and Licensed Know-How will revert to Harvard and neither Licensee nor its Affiliates may make any further use or exploitation of the Licensed Patent Rights or Licensed Know-How; and (b) any existing agreements that contain a Sublicense shall terminate to the extent of such Sublicense; provided, however, that, for each Sublicensee, upon termination of the Sublicense agreement with such Sublicensee, if the Sublicensee is not then in breach of its Sublicense agreement with Licensee such that Licensee would have the right to terminate such Sublicense, such Sublicensee shall have the right to seek a license from Harvard. Harvard agrees to negotiate such licenses in good faith under reasonable terms and conditions, which shall not impose any representations, warranties, obligations or liabilities on Harvard that are not included in this Agreement.
- **10.3.2** Accruing Obligations. Termination or expiration of this Agreement shall not relieve the Parties of obligations accruing prior to such termination or expiration, including obligations to pay amounts accruing hereunder up to the date of termination or expiration, which shall be paid to Harvard within forty-five (45) days after such termination or expiration. Notwithstanding any other provision of this Agreement, after the date of termination or expiration (except in the case of termination by Harvard pursuant to Section 10.2), Licensee, its Affiliates and Sublicensees (a) may sell or otherwise dispose of all Licensed Products then in stock and (b) may complete the production of Licensed Products then in the process of production and sell or otherwise dispose of the same, in each case, in accordance with the terms and conditions of this Agreement, including that in the case of both (a) and (b), Licensee shall pay the applicable royalties and payments to Harvard in accordance with Article 4 within thirty (30) days after the end of each Calendar Quarter after the Term during which such sales or other dispositions are made, provide reports and audit rights to Harvard pursuant to Article 5 and maintain insurance in accordance with the requirements of Section 9.2.
- **10.4 Survival.** The Parties' respective rights, obligations and duties under <u>Articles 5, 9, 10</u> and <u>11</u> and <u>Sections 8.2</u> and <u>8.3</u>, as well as any rights, obligations and duties which by their nature extend beyond the expiration or termination of this Agreement, shall survive any expiration or termination of the Agreement shall survive such expiration or termination.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

11. Miscellaneous.

- 11.1 Preference for United States Industry. During the period of exclusivity of this license in the United States, Licensee shall comply with 37 C.F.R. § 401.14(i) or any successor rule or regulation. Upon Licensee's request, and at Licensee's expense, Harvard shall apply to the applicable United States governmental agency for a waiver to such requirements; provided, however, that all reasonable costs incurred by Harvard related to the preparation and application of the waiver shall be paid by Licensee within forty-five (45) days following receipt of Harvard's valid and accurate invoice for such costs. In the event that Harvard is unable to obtain such waiver, then subject to the Parties mutually agreeing upon reduced royalty rates and other appropriate modifications to this Agreement to address Licensee's loss of exclusivity, the license granted to Licensee under Section 2.1 shall become non-exclusive.
- 11.2 **No Security Interest**. Licensee shall not enter into any agreement under which Licensee grants to or otherwise creates in any third party a security interest in this Agreement or any of the rights granted to Licensee herein. Any grant or creation of a security interest purported or attempted to be made in violation of the terms of this Section 11.2 shall be null and void and of no legal effect.
- 11.3 Use of Name. Except as provided below, Licensee shall not, and shall ensure that its Affiliates and Sublicensees shall not, use or register the name "Harvard" or "Wyss Institute" (alone or as part of another name) or any logos, seals, insignia or other words, names, symbols or other indicia of origin owned by Harvard that, in each case, identify Harvard or the Wyss Institute or any other Harvard school, unit, division or affiliate (collectively, "Harvard Names") for any purpose as a trademark except with the prior written approval of, and in accordance with restrictions required by, Harvard, provided, however that Company may issue a press release substantially in the form set forth in Exhibit F to the Collaboration Agreement, which press release has been approved in advance by Harvard. This restriction shall not apply to any information required by law to be disclosed to any Governmental Authority, including, without limitation, to satisfy any securities reporting requirements. If, notwithstanding this prohibition, Licensee registers any Harvard Name as a trademark, service mark, domain name, trade name, business or company name or otherwise anywhere in the world, then, in addition to any other remedies Harvard may have, Harvard shall have the right to compel Licensee to assign Licensee rights in such registration to Harvard and Licensee shall take such steps as may be necessary to transfer record ownership of such registration to Harvard, at Licensee cost.
- **11.4 Entire Agreement.** This Agreement is the sole agreement with respect to the subject matter hereof and except as expressly set forth herein, supersedes all other agreements and understandings between the Parties with respect to the same.
- 11.5 **Notices.** Unless otherwise specifically provided, all notices required or permitted by this Agreement shall be in writing and may be delivered personally, or may be sent by facsimile (if applicable), expedited delivery or certified mail, return receipt requested, to the following addresses or by email (with confirmatory copy sent by regular mail), unless the Parties are subsequently notified of any change of address in accordance with this Section 11.5:

If to Licensee: ReWalk Robotics, Ltd. 200 Donald Lynch Blyd.

Marlborough, Massachusetts 01752

Email: <u>larry.jasinski@rewalk.com</u> and

ofir@rewalk.com

Facsimile: +1 (508) 251-2970 Attn.: Larry Jasinski and Ofir Koren

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

With copy to the following (if a

technical notice):

ReWalk Robotics, Ltd. Hatnufa st. 3, floor 6

Yokneaam, POB 161, 2069203

Israel

Facsimile: +972-4-959 0125

Email: larry.jasinski@rewalk.com and

ofir@rewalk.com

Attn.: Larry Jasinski and Ofir Koren

If to Harvard: Office of Technology Development

Harvard University

Richard A. and Susan F. Smith Campus Center, Suite 727

1350 Massachusetts Avenue Cambridge, Massachusetts 02138 Facsimile: +1 (617) 495-9568

Attn.: Chief Technology Development Officer

Any notice shall be deemed to have been received as follows: (a) by personal delivery or expedited delivery, upon receipt; (b) by facsimile, one business day after transmission or dispatch; (c) by certified mail, as evidenced by the return receipt. If notice is sent by facsimile, a confirming copy of the same shall be sent by mail to the same address.

- 11.6 Governing Law and Jurisdiction. This Agreement will be governed by, and construed in accordance with, the substantive laws of the Commonwealth of Massachusetts, without giving effect to any choice or conflict of law provision, except that questions affecting the construction and effect of any patent shall be determined by the law of the country in which the patent shall have been granted. Any action, suit or other proceeding arising under or relating to this Agreement, its subject matter, any document or instrument delivered pursuant to, or a breach of this Agreement or any such document or instrument (a "Suit") shall be brought in a court of competent jurisdiction in the Commonwealth of Massachusetts, and the Parties hereby consent to the sole jurisdiction of the state and federal courts sitting in the Commonwealth of Massachusetts. Each Party agrees not to raise any objection at any time to the laying or maintaining of the venue of any Suit in any of the specified courts, irrevocably waives any claim that Suit has been brought in any inconvenient forum and further irrevocably waives the right to object, with respect to any Suit, that such court does not have any jurisdiction over such Party.
- **11.7 Binding Effect.** This Agreement shall be binding upon and inure to the benefit of the Parties and their respective legal representatives, successors and permitted assigns.
 - **11.8 Headings.** Section and subsection headings are inserted for convenience of reference only and do not form a part of this Agreement.
- 11.9 Counterparts. The Parties may execute this Agreement in two or more counterparts, each of which shall be deemed an original, but both of which together shall constitute one and the same instrument. Transmission by facsimile or electronic mail of an executed counterpart of this Agreement shall be deemed to constitute due and sufficient delivery of such counterpart. If by electronic mail, the executed Agreement must be delivered in a .pdf format.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- **11.10 Amendment; Waiver**. This Agreement may be amended, modified, superseded or cancelled, and any of the terms may be waived, only by a written instrument executed by each Party or, in the case of waiver, by the Party waiving compliance. The delay or failure of either Party at any time or times to require performance of any provisions hereof shall in no manner affect the rights at a later time to enforce the same. No waiver by either Party of any condition or of the breach of any term contained in this Agreement, whether by conduct, or otherwise, in any one or more instances, shall be deemed to be, or considered as, a further or continuing waiver of any such condition or of the breach of such term or any other term of this Agreement.
- 11.11 No Agency or Partnership. Nothing contained in this Agreement shall give either Party the right to bind the other, be deemed to constitute either Party as agent for or partner of the other or any third party, or be construed or fiduciary relationship between the Parties. The relationship between the Parties is that of independent contractors.

11.12 Assignment and Successors.

- 11.12.1 This Agreement may not be assigned by either Party without the consent of the other, which consent shall not be unreasonably withheld, except that each Party may, without such consent, assign this Agreement, in its entirety, and the rights, obligations and interests of such Party (a) to any affiliate or (b) to any purchaser of all of such Party's equity or all or substantially all of such Party's assets or business to which this Agreement relates, or (c) to any successor corporation resulting from any merger or consolidation of such Party with or into such corporation, provided, in each case, that the assignee agrees in writing to be bound by the terms of this Agreement. Any assignment purported or attempted to be made in violation of the terms of this Section 11.12 shall be null and void and of no legal effect.
- 11.12.2 The rights and licenses granted to Licensee hereunder shall run with the Licensed Patent Rights and Licensed Know-How and Harvard shall not license, encumber, convey or otherwise transfer its rights in or to the Licensed Patent Rights or Licensed Know-How in any way that conflicts with the terms of this Agreement or the rights and licenses granted to Licensee hereunder without the prior written consent of Licensee (in its sole discretion). Any assignment, conveyance, license, encumbrance, transfer or other disposition of Harvard's rights in or to the Licensed Patent Rights or Licensed Know-How shall be expressly subject to this Agreement and the rights and licenses granted to Licensee hereunder.
- **11.13 Force Majeure**. Except for monetary obligations hereunder, neither Party will be responsible for delays resulting from causes beyond the reasonable control of such Party, including fire, explosion, flood, war, strike, or riot, provided that the nonperforming Party uses commercially reasonable efforts to avoid or remove such causes of nonperformance and continues performance under this Agreement with reasonable dispatch whenever such causes are removed.
- **11.14 Interpretation**. Each Party hereto acknowledges and agrees that: (a) it and/or its counsel reviewed and negotiated the terms and provisions of this Agreement and has contributed to its revision; (b) the rule of construction to the effect that any ambiguities are resolved against the drafting Party shall not be employed in the interpretation of this Agreement; (c) the terms and provisions of this Agreement shall be construed fairly as to both Parties hereto and not in favor of or against either Party, regardless of which Party was generally responsible for the preparation of this Agreement; and (d) the use of "include," "includes," or "including" herein shall not be limiting and "or" shall not be exclusive.
- **11.15 Severability.** If any provision of this Agreement is or becomes invalid or is ruled invalid by any court of competent jurisdiction or is deemed unenforceable, such invalidity shall not render the entire Agreement unenforceable or invalid but rather the Agreement shall be read and construed as if the invalid or unenforceable provision(s) are not contained herein, and the rights and obligations of the Parties shall be construed and enforced accordingly.

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

- 11.16 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the Parties hereto and their respective successors and permitted assigns and nothing herein, express or implied, is intended to or shall confer upon any other individual, corporation, partnership, joint venture, limited liability company, Governmental Authority, unincorporated organization, trust, association or other entity any legal or equitable right, benefit or remedy of any nature whatsoever, under or by reason of this Agreement.
- **11.17 Reimbursement**. To the extent that Licensee reimburses Harvard for any costs incurred by Harvard for the preparation, filing, prosecution and maintenance of any Patent Rights hereunder (such costs, "**Reimbursed Costs**"), then if Harvard grants a license under such Patent Rights to a third Person outside the Field (each, an "**Ex-Field Licensee**") and receives reimbursement from such Ex-Field Licensee for Reimbursed Costs, then Licensee shall only be responsible for reimbursing an appropriate portion of such Reimbursed Costs based on the scope and type of license (i.e., non-exclusive vs. exclusive) granted to such Ex-Field Licensee and Harvard shall credit Company's portion of the Reimbursed Costs to amounts owed to Harvard hereunder.
- Licenses of Intellectual Property. Each Party acknowledges and agrees that all rights and licenses under the Licensed Patent Rights and Licensed Know-How granted by Harvard to Licensee under this Agreement are, and shall otherwise be deemed to be, for purposes of the United States Bankruptcy Code (the "Bankruptcy Code"), including Section 365(n) thereof, licenses of rights to "intellectual property" as defined under Section 101(35A) of the Bankruptcy Code and in connection therewith, Section 365(n) shall apply. The Parties acknowledge and agree that Licensee shall retain and may fully exercise all of its respective rights and elections under the Bankruptcy Code, including, without limitation, Licensee's election to continue receiving the benefit of the licenses granted hereunder in consideration for the royalties due and payable hereunder. Without limiting the foregoing, if Harvard makes a general assignment for the benefit of creditors, if a receiver for Harvard's assets is appointed, if Harvard becomes insolvent, makes a filing in bankruptcy or ceases operations, or if any of Harvard's lenders forecloses upon or otherwise exercises any of its rights under or with respect to any liens or security interests on any of the Licensed Patent Rights, Licensed Know-How and/or this Agreement, then Harvard acknowledges and agrees that such foreclosure and exercise shall be expressly subject to this Agreement and all rights and licenses under the Licensed Patent Rights and Licensed Know-How granted by Harvard to Licensee under this Agreement and the license granted to Licensee hereunder shall, in consideration for the royalties due and payable hereunder, continue in full force and effect.
- 11.19 Confidentiality. The terms (but not the existence) of this Agreement constitute the Confidential Information of both Parties pursuant to Article VII of the Collaboration Agreement and neither Party shall disclose the terms of this Agreement to any third party except (a) to its Affiliates, Distributors and Sublicensees and their respective officers, directors, members, employees, agents and outside advisors who reasonably need to know such information for exercising such Party's rights or performing such Party's obligations under this Agreement or the Collaboration Agreement, (b) with the prior written approval of the other Party, (c) the Company may issue a press release substantially in the form set forth in Exhibit F to the Collaboration Agreement, which press release has been approved in advance by Harvard, or (d) this restriction shall not apply to any information required by law to be disclosed to any Governmental Authority or required to be disclosed publicly pursuant to applicable law, including, without limitation, pursuant to or in connection with securities law reporting obligations or the rules of any securities exchange, provided that the Party disclosing such information shall seek confidential treatment of such Confidential Information to the extent permitted under applicable law.

[Signature page follows]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed by their duly authorized representatives as of the date first written above.

Presid	ent and Fellows of Harvard College	ReWal	k Robotics, Ltd.
By:	/s/ Isaac T. Kohlberg	By:	/s/ Larry Jasinski
Name:	Isaac T. Kohlberg	Name:	Larry Jasinski
	Senior Associate Provost		· · ·
	Chief Technology Development Officer		
	Office of Technology Development		
Title:	Harvard University	Title:	CEO
	ndersigned, hereby confirm that I have read the Agreement, that its cong the provisions of <u>Section 2.6</u> .	ontents are acc	reptable to me and that I will act in accordance with its terms,
/s/ Con	or Walsh		
Conor	Walsh, Ph.D.		
	24	-	
	ential treatment has been requested for redacted portions of this exhibit. This copy omits th of this exhibit has been provided separately to the Securities and Exchange Commission.	e information subj	ect to the confidentiality request. Omissions are designated as [**]. A complete

Exhibit 1.11 **Development Milestones**

Development Milestone		Deadline
•	[**]	On or before [**].
•	[**]	On or before [**].
•	[**]	On or before [**].
•	[**]	On or before [**].
•	[**]	On or before [**].
•	[**]	On or before [**].

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit 1.12 **Development Plan**

[**].

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit 1.24 Licensed Know-How

* Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Exhibit 1.25

Licensed Patent Rights

Harvard Background Patents

Case	Country	Туре	Appl. Title	Serial Number
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.

Case	Country	Туре	Appl. Title	Serial Number
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]
[**]	[**]	[**]	[**]	[**]

Licensed Type A Patent Rights

[**]

Consulting Patent Rights

**]

Joint Consulting Patent Rights

[**]

^{*} Confidential treatment has been requested for redacted portions of this exhibit. This copy omits the information subject to the confidentiality request. Omissions are designated as [**]. A complete version of this exhibit has been provided separately to the Securities and Exchange Commission.



FOR IMMEDIATE RELEASE May 17, 2016

ReWalk Announces Collaboration with Harvard University's Wyss Institute to Develop Lightweight and Soft Exoskeleton Systems for the Treatment of Stroke, Multiple Sclerosis and Limited Mobility Patients

Five-Year Agreement Establishes Unique Collaboration and Extensive IP Licensing for "Soft Suit" Exoskeleton Systems

YOKNEAM ILIT, Israel and MARLBOROUGH, MA — ReWalk Robotics Ltd. (Nasdaq: RWLK) ("ReWalk" or "Company"), the leading global exoskeleton developer and manufacturer today announced a collaboration with Harvard University's Wyss Institute for Biologically Inspired Engineering for the licensing of Intellectual Property (IP), and development of concepts and designs of lightweight exoskeleton system technologies for lower limb disabilities. This exclusive licensing and collaboration agreement will focus on the development of "soft suit" systems for the treatment of stroke, Multiple Sclerosis (MS), mobility limitations for the elderly and other medical applications.

"There is a great need in the health care system for lightweight, lower-cost wearable exoskeleton designs to support stroke patients, individuals diagnosed with Multiple Sclerosis and senior citizens who require mechanical mobility assistance. This collaboration will help create the next generation of exoskeleton systems, making life-changing technology available to millions of consumers across a host of patient populations," said Larry Jasinski, CEO of ReWalk.

"This is a very exciting day for the soft suit technology," said Conor Walsh, who is a Core Faculty Member at the Wyss Institute, the John L. Loeb Associate Professor of Engineering and Applied Sciences at Harvard's John A. Paulson School of Engineering and Applied Sciences and Founder of the Harvard Biodesign Lab. "ReWalk brings commercialization expertise and experience in the area of wearable robotics and complements our translation-focused research. Ultimately this deal paves the way for this technology to make its way to patients."

The majority of Stroke and MS patients as well as the elderly do not require the structural support of current market rigid exoskeleton systems for individuals with spinal cord injury. The soft suit prototypes from the Wyss Institute transmit power to key joints of the legs with cable technologies powered with software and mechanics that are similar to the technologies used in the ReWalk system. The cables are connected to fabric-based designs that attach to the legs and foot, thus lending the name "soft suit."

"Our collaboration with ReWalk is a wonderful example of the Wyss Institute model in action," said Wyss Institute Founding Director Don Ingber, M.D., Ph.D. "We work with industry to help de-risk the technologies we develop, both technically and commercially, and thereby expedite their translation into real world applications."

Initial pilot studies with stroke patients run at Harvard's Wyss Institute in collaboration with faculty and researchers from Boston University have demonstrated the function of the soft suit exoskeleton technology. ReWalk will work in concert with the Wyss Institute on the continued development of lightweight designs to complete clinical studies, pursue regulatory approvals and commercialize the systems on a global scale. The first commercial application is expected to be stroke, followed by MS and then additional applications. There are an estimated 3 million stroke patients with lower limb disability in the U.S., and approximately 400,000 individuals with Multiple Sclerosis.

"Harvard and its Wyss Institute are pioneers in the development of technology in this space. The licensed Harvard patent portfolio currently includes 19 patent applications, which includes applications in at least six countries. The applications cover the soft suit, control systems and methods of treating patients. Harvard and the Wyss Institute have built comprehensive research expertise in addition to the worldwide patent portfolio. There is no better partner than these renowned institutions with which to pursue the mission of bringing cutting-edge technology to disabled individuals around the world," Jasinski added.

Coordinated by Harvard's Office of Technology Development, this collaboration includes funding for continued research and technology development at the Wyss Institute and transfer of knowledge and research results to ReWalk.

The agreement is effective May 16, 2016, with anticipated commercialization before 2019.

For more about Harvard's Wyss Institute, please visit: http://wyss.harvard.edu/

For more about Harvard's Office of Technology Development, please visit: http://otd.harvard.edu/

For more about ReWalk, please visit: www.rewalk.com

About ReWalk Personal 6.0

ReWalk Personal 6.0 is a wearable robotic exoskeleton that provides powered hip and knee motion to enable individuals with spinal cord injury to stand upright and walk. The system provides user-initiated mobility through the integration of a wearable brace support, a computer-based control system and motion sensors. The system allows independent, controlled walking similar to a natural gait pattern of the legs. The ReWalk device is the most studied exoskeleton in the industry

About ReWalk Robotics Ltd.

ReWalk Robotics Ltd. develops, manufactures and markets wearable robotic exoskeletons for individuals with spinal cord injury. Our mission is to fundamentally change the quality of life for individuals with lower limb disability through the creation and development of market leading robotic technologies. Founded in 2001, ReWalk has headquarters in the U.S., Israel and Germany. For more information on the ReWalk systems, please visit http://www.rewalk.com.

ReWalk® is a registered trademark of ReWalk Robotics Ltd. in Israel.

Forward Looking Statements

In addition to historical information, this press release contains forward-looking statements within the meaning of the U.S. Private Securities Litigation Reform Act of 1995, Section 27A of the U.S. Securities Act of 1933, and Section 21E of the U.S. Securities Exchange Act of 1934. Such forward-looking statements may include projections regarding ReWalk's future performance and, in some cases, may be identified by words like "anticipate," "assume," "believe," "continue," "could," "estimate," "expect," "intend," "may," "plan," "potential," "predict," "project," "future," "will," "seek" and similar terms or phrases. The forward-looking statements contained in this press release are based on management's current expectations, which are subject to uncertainty, risks and changes in circumstances that are difficult to predict and many of which are outside of ReWalk's control. Important factors that could cause ReWalk's actual results to differ materially from those indicated in the forward-looking statements include, among others: ReWalk's expectations regarding future growth, including its ability to increase sales in its existing geographic markets and to expand to new markets; ReWalk's ability to maintain and grow its reputation and to achieve and maintain market acceptance of its products; ReWalk's ability to achieve reimbursement from third-party payors for its products; ReWalk's expectations as to its clinical research program and clinical results; ReWalk's ability to improve its products, develop new products; ReWalk's ability to maintain adequate protection of its intellectual property and to avoid violation of the intellectual property rights of others; ReWalk's ability to repay its secured indebtedness; ReWalk's ability to gain and maintain regulatory approvals; ReWalk's ability to maintain relationships with existing customers and develop relationships with new customers; and other factors discussed under the heading "Risk Factors" in ReWalk's Annual Report on Form 10-K for the year ended December 31, 2015 filed with the U.S. Securities and Exchange Commission on February 29, 2016 and other documents subsequently filed with or furnished to the U.S. Securities and Exchange Commission. Any forward-looking statement made in this press release speaks only as of the date hereof. Factors or events that could cause ReWalk's actual results to differ from the statements contained herein may emerge from time to time, and it is not possible for ReWalk to predict all of them. Except as required by law, ReWalk undertakes no obligation to publicly update any forward-looking statements, whether as a result of new information, future developments or otherwise.

Media Contact:

ReWalk Robotics Jennifer Wlach 202-261-4000 jwlach@mercuryllc.com

Wyss Institute at Harvard University Mary Tolikas 617-432-7733 mary.tolikas@wyss.harvard.edu

Investor Relations Contact: Lisa M. Wilson In-Site Communications, Inc. 917-543-9932 lwilson@insitecony.com